MEMORANDUM OF INCORPORATION

of

MONAGHAN FARM NPC

which is a non-profit company, with not less than 7 (seven) Directors

REGISTRATION NUMBER: 2008/001430/08

REGISTRATION DATE: 23 JANUARY 2008

as required in terms of

THE COMPANIES ACT, NO 71 OF 2008
(AS AMENDED)

TABLE OF CONTENTS

Memorandun	n of Incorporation	
1		4
2	INCORPORATION	11
3		13
	POWERS OF THE ASSOCIATION	
4	20020072 02020.0	13
	OPTIONAL PROVISIONS OF THE COMPANIES ACT	14
	SPECIAL CONDITIONS	14
	RESOLUTIONS	15
_	FINANCIAL YEAR	15
	ACCOUNTING RECORDS AND FINANCIAL STATEMENTS	15
	MEMBERSHIP AND CLASSES OF MEMBERS	16
	LEVIES AND CONTRIBUTIONS	19
	AMENDMENTS TO THE MOI	26
	THE PROCEDURE FOR MAKING OF RULES	27
	REGULATIONS, & HOUSE RULES	28
	PUBLICATION OF RULES, REGULATIONS, & HOUSE RULES	29
_	ENFORCEMENT & DISPUTE	30
	MEMBERS' MEETINGS	31
	NOTICE OF MEMBERS' MEETING	33
	CHAIR, QUORUM & VOTING PROXIES	34 37
20 21		38
	DIRECTORS, ELECTION OF DIRECTORS AND CASUAL	39
22	VACANCIES	39
23	CESSATION OF OFFICE AS DIRECTOR	41
	REMUNERATION OF DIRECTORS AND MEMBERS OF BOARD	42
24	COMMITTEES	72
25	FINANCIAL ASSISTANCE FOR DIRECTORS AND PRESCRIBED	42
20	OFFICERS AND THEIR RELATED AND INTER-RELATED PARTIES	12
26	GENERAL POWERS AND DUTIES OF DIRECTORS AND	43
20	ELECTION OF CHAIRMAN AND VICE CHAIRMAN	. •
27	BOARD COMMITTEES	45
	PERSONAL FINANCIAL INTERESTS OF DIRECTORS	45
	PROCEEDINGS OF DIRECTORS	47
30		50
	COMMON PROPERTY	
31	ALIENATION	50
32	LOSS OF DOCUMENTS	51
33	NOTICES	51
34	INDEMNITY	52
35	ACCEPTANCE OF CESSIONS OF RIGHTS	54
36	ACCEPTANCE OF COMMON FACILITIES	54
Annexure 1	Definitions in the Companies Act	
Annexure 2	Directors - Ineligible / disqualified in terms of Section 69(7) and (8) of	
	the Companies Act read with Regulation 39(3)	
Annexure 3 Annexure 4	Prescribed Methods of Delivery in the Regulations	

Attached

RULES

REGULATIONS

SCHEDUEL OF FEES AND PENALTIES

1. INTERPRETATION AND DEFINITIONS

- 1.1. In this MOI, unless the context clearly indicates a contrary intention, the following words and expressions bear the meanings assigned to them and cognate expressions bear corresponding meanings —
 - 1.1.1. "Act" means the Companies Act, No. 71 of 2008, as amended from time to time;
 - 1.1.2. "alienate" means the transfer of any right or interest whatsoever in nature in respect of any Erf, unit, part thereof or undivided share therein howsoever arising and whether voluntary or involuntary and includes by way of sale, exchange, donation, deed, intestacy, will, cession, assignment, court order of insolvency, irrespective of whether such alienation is subject to a suspensive or resolutive condition and alienating has a corresponding meaning;
 - 1.1.3. "Article" means each separately numbered paragraph of this MOI;
 - 1.1.4. **"Board"** means the board of Directors from time to time of the Association or if there is only one Director, then that Director;
 - 1.1.5. **"Business Day"** means any day other than a Saturday, Sunday or public holiday in the Republic;
 - 1.1.6. **"Chairman"** means the Chairman elected in terms of Article 20.2;
 - 1.1.7. **"Commission"** means the Companies and Intellectual Property Commission established by section 185 of the Act;
 - 1.1.8. **"Company"** means Monaghan Farm NPC, with registration number 2008/001430/07, herein referred to as "the Association";
 - 1.1.9. "common facilities" or "common amenities" is a reference to the organic farm facility, stables and equestrian facilities, clubhouse, gymnasium, sports facilities including four tennis courts and a squash court, a private pre-school facility, perimeter security, gatehouse and internal roads sewerage plant, refuse yard and all areas referred to as private open space in the Conditions of Establishment for the Township of Monaghan and all extensions thereof;

- 1.1.10. "common property" means such property of the Association that may be used by all Members or registered in the name of the Association such as the common areas, facilities, servitudes and open spaces on Monaghan Farm;
- 1.1.11. **"Conditions of Establishment"** means those conditions of establishment issued by the relevant authorities governing the erven on Monaghan Farm.
- 1.1.12. **"CPI"** means the annual rate of consumer price inflation as published from time to time by Nedbank;
- 1.1.13. **"Deeds Office"** means the office of the Registrar of Deeds in which jurisdiction ownership of land within Monaghan Farm, is registered;
- 1.1.14. "Deeds Registries Act" means the Deeds Registries Act 47 of 1937;
- 1.1.15. **"Developer"** means Clewer Development Trust (Registration number IT 4830/2006) or its successors to a right or rights to develop the Property;
- 1.1.16. **"development"** means, together, the residential and other elements developed, or to be developed, on Monaghan Farm,;
- 1.1.17. **"development period"** means the period from incorporation of the Company up to the date upon which the Developer ceases to be a registered owner of any land within Monaghan Farm, or 28 February 2019, whichever is the earlier date;
- 1.1.18. "Director" means a member of the Board as contemplated in section 66 of the Act, and includes any person occupying the position of a director, by whatever name designated;
- 1.1.19. "Electronic Communication" has the meaning set out in Section 1 of the Electronic Communication and Transactions Act, Act 25 of 2002, and includes a communication by means of a data message consisting of data generated, sent, received or stored by electronic means including electronic mail (e-mail), fax, SMS and, where applicable, Skype or any other video conferencing communication or electronic virtual platform communication available.;
- 1.1.20. **"EMP"** means the Environmental Management Plan from time to time in force and effect applicable to the development and Monaghan Farm;.
- 1.1.21. **"Encumbrance"** means any right of first refusal, purchase right, option or any other restriction of any kind on ownership, transfer, use, possession, receipt of

- income from or any other exercise of any attribute of ownership, including any mortgage, pledge, lien or other security interest;
- 1.1.22. "Erf" means any Erf in Monaghan Farm upon which has been, or is to be, erected any house or dwelling, sectional title or residential units, or is being used as common property, and which Erf is capable of registration in the Deeds Office and includes, where practical and unless the context clearly requires the contrary, any subdivision;
- 1.1.23. "facilities" means any facilities of whatsoever nature which have been or may be provided within or on Monaghan Farm including without limitation the amenities, common facilities and such other facilities relating to the entertainment, recreation or otherwise, erected on the common property, the use of which may be exclusive, reserved or conditional;
- 1.1.24. "future erven" means the residential erven, or erf, established in terms of the applicable town planning scheme and situant within that north-east quadrant of Monaghan Farm which is bounded to its west and south by the Jukskei River;
- 1.1.25. **"improvements"** in respect of land shall include the erection of a house, dwelling or any other structure;
- 1.1.26. **"Income Tax Act"** means the Income Tax Act 58 of 1962, as amended from time to time;
- 1.1.27. "land" means any Erf on Monaghan Farm, or any approved subdivided portion thereof, including units in any sectional title developments, with or without improvements, excluding the streets;
- 1.1.28. **"levy contribution"** means the contribution levied upon members, as determined from time to time pursuant to Article 11;
- 1.1.29. **"local authority"** means the Local Municipality in whose municipal boundaries Monaghan Farm falls;
- 1.1.30. "Member" means a member of the Association and reflected in the records of the Deeds Office as the registered owner of land or of an undivided share in land within Monaghan Farm;
- 1.1.31. "Members' Meeting" means any meeting of the Members and refers collectively to the Annual General Meeting and any Special General Meetings of the Association;

- 1.1.32. "Manager" means the manager of Monaghan Farm appointed from time to time;
- 1.1.33. **"MOI"** means this Memorandum of Incorporation together with any such amendments thereto from time to time;
- 1.1.34. **"Monaghan Farm"** means the residential estate on the Property including all facilities, amenities and approved or to be approved activities conducted;
- 1.1.35. "month" means a calendar month;
- 1.1.36. **"office"** means the registered office of the Association;
- 1.1.37. "profit" includes revenue and capital profits;
- 1.1.38. "Property" means Portion 3 (A portion of portion 2) of the Farm Vlakfontein 494 JQ, as it existed prior to proclamation thereon and shall include the townships proclaimed, or to be proclaimed, thereon;
- 1.1.39. "purchaser" means any person who has acquired rights, regardless as to the nature therein relative to and which gives rise or purports to give rise to an entitlement to occupy land or an undivided share therein forming part of Monaghan Farm;
- 1.1.40. **"Regulations"** means the Regulations published in terms of the Act from time to time and any reference to a single Regulation shall be a reference to the corresponding Regulation in the Regulations;
- 1.1.41. "Republic" means the Republic of South Africa;
- 1.1.42. "scheme" means any sectional title scheme or development scheme having a fraction assigned to each unit of the scheme, established in terms of the Sectional Titles Act;
- 1.1.43. **"Section"** means the corresponding section in the Act, or if the section is qualified by a reference to a specific act, the corresponding section in that act;
- 1.1.44. "Sectional Titles Act" means the Sectional Titles Act 95 of 1986, as amended;
- 1.1.45. **"services"** means the provisions of security, road maintenance, common property facilities and such other utilities and amenities as may be provided to Members at the instance of or with approval of the Board;

- 1.1.46. **"Solvency and Liquidity Test"** has the meaning attributed thereto in section 4 of the Act;
- 1.1.47. **"streets"** means the roads that have been constructed on Monaghan Farm including all right of way servitudes;
- 1.1.48. **"subdivision"** means any subdivision of an Erf and includes any reference to "a subdivided Erf", a "subdivision of an Erf" etc.;
- 1.1.49. **"tenancy agreement"** means any agreement in respect of any common facility for its lease, tenancy, management, operation or similar purpose
- 1.1.50. **"town planning scheme"** means the operative town planning scheme applicable to Monaghan Farm and registered by the relevant local authority;
- 1.1.51. **"township"** means a township proclaimed in terms of the provisions of the Town and Town-planning Ordinance, Transvaal, 15 of 1986, or any replacement legislation;
- 1.1.52. **"unit"** means a unit in a sectional title scheme (inclusive of the exclusive use areas) with or without outbuildings and situated on its own residential Erf or individual subdivision, tenure or which may be registered in the Deeds Offices.
- 1.2. In this MOI, unless the context clearly indicates otherwise
 - 1.2.1. words and expressions defined in the Act and which are not defined herein shall have the meanings given to them in the Act;
 - 1.2.2. a reference to a section of the Act by number refers to the corresponding section of the Act notwithstanding the renumbering of such section after the date on which this MOI is registered;
 - 1.2.3. in any instance where there is a conflict between a provision (be it expressed, implied or tacit) of this MOI and
 - 1.2.3.1. an alterable or elective provision of the Act, the provision of this MOI shall prevail to the extent of the conflict; and
 - 1.2.3.2. an unalterable or non-elective provision of the Act, the unalterable or non-elective provision of the Act shall prevail to the extent of the conflict, unless the MOI imposes on the Association a higher standard, greater restriction, longer period of time or similarly more onerous

requirement, in which event the relevant provision of this MOI shall prevail to the extent of the conflict;

- 1.2.4. article headings are for convenience only and are not to be used in its interpretation;
- 1.2.5. an expression which denotes -
 - 1.2.5.1. any gender includes the other genders;
 - 1.2.5.2. a natural person includes a juristic person and vice versa; and
 - 1.2.5.3. the singular includes the plural and vice versa.
- 1.2.6. if the due date for performance of any obligation in terms of this MOI is a day which is not a Business Day then (unless otherwise stipulated), the due date for performance of the relevant obligation shall be the immediate succeeding Business Day;
- 1.2.7. any reference to a notice shall be construed as a reference to a written notice, and shall include a notice which is transmitted electronically in a manner and form permitted in terms of the Act and/or the Regulations.
- 1.3. Any reference in this MOI to -
 - 1.3.1. "days" shall be construed as calendar days unless qualified by the word "business" and when this MOI provides for the happening of one event and another, the number of days must be calculated by:
 - 1.3.1.1. excluding the day on which the first such event occurs;
 - 1.3.1.2. including the day on or by which the second event is to occur;
 - and, in respect of business days only, shall be adjusted by:
 - 1.3.1.3. excluding any public holiday, Saturday or Sunday that falls on or between the days contemplated in Articles 1.3.1.1 and 1.3.1.2, respectively; and
 - 1.3.1.4. if that day is not a Business Day, then same shall be the first succeeding Business Day;
 - 1.3.2. **"law"** means any law of general application and includes the common law and any statute, constitution, decree, treaty, regulation, directive, ordinance, by-law,

order or any other enactment of legislative measure of government (including local and provincial government) statutory or regulatory body which has the force of law and a reference to any statutory enactment shall be construed as a reference to that enactment as amended or substituted from time to time;

- 1.3.3. "writing" means legible writing and in English and includes printing, typewriting, lithography or any other mechanical process, as well as any electronic communication in a manner and a form permitted in terms of the Act and/or the Regulations.
- 1.4. The words "include" and "including" mean "include without limitation" and "including without limitation". The use of the words "include" and "including" followed by a specific example or examples shall not be construed as limiting the meaning of the general wording preceding it.
- 1.5. Unless otherwise provided in this MOI or the Act:
 - 1.5.1. Defined terms appearing herein in title case shall be given their meaning as defined, while the same terms appearing in lower case shall (except where defined in the Act) be interpreted in accordance with the plain English meaning; and
 - 1.5.2. Any words and expressions defined in any Article shall, unless the application of such word or expression is specifically limited to that Article, bear the meaning assigned to such word.
- 1.6. Where figures are referred to in numerals and in words, and there is any conflict between the two, the words shall prevail, unless the context indicates a contrary intention.
- 1.7. Any reference herein to "this MOI" shall be construed as a reference to this MOI as amended from time to time.
- 1.8. Any Schedules attached to this MOI form part of this MOI.
- 1.9. The standard form of memorandum of incorporation for a Non-Profit Company with Members contained in Annexure 1 of the Regulations to the Act shall not apply to the Association.
- 1.10. This MOI is in a form unique to the Association, as contemplated in section 13(1)(a)(ii) of the Act.
- 1.11. This MOI, and any rules of the Association, are binding –

- 1.11.1. between the Association and each Member;
- 1.11.2. between or among the Members; and
- 1.11.3. between the Association and -
 - 1.11.3.1. each Director or prescribed officer of the Association; or
 - 1.11.3.2. any other person serving the Association as a member of a committee of the Board;

in the exercise of their respective functions within the Association.

2. INCORPORATION

The Association -

- 2.1. is a Non-Profit Company with voting members, incorporated for a public benefit or other object as required by Item 1(3) of Schedule 1 of the Act;
- 2.2. cannot distribute its income, profits and property to its Incorporators, Members, Directors, Officers or persons related to any of them, except to the extent permitted by Item 1(3) of Schedule 1 to the Companies Act.
- 2.3. The Association is incorporated in accordance with and governed by
 - 2.3.1. the unalterable provisions of the Act, save to the extent that this MOI imposes on the Company a higher standard, greater restriction, longer period of time or similarly more onerous requirement; and
 - 2.3.2. the alterable provisions of the Act, subject to the limitations, extensions, variations or substitutions set out in this MOI; and
 - 2.3.3. the other provisions of this MOI.
- 2.4. It is recorded that the Association may adopt and shall be governed by the following which are recorded in order of precedence:
 - 2.4.1. Rules, formulated in accordance with this MOI, and filed at the Commission; and
 - 2.4.2. Regulations, for the purposes of advancing the objectives of the Association, not requiring filing with the Commission; and
 - 2.4.3. House Rules, not requiring filing with the Commission.

3. POWERS OF THE ASSOCIATION

- 3.1. The Association has all of the legal powers and capacity of an individual, except to the extent that a juristic person is incapable of exercising any such power, or having any such capacity or this MOI otherwise limits such legal powers and capacity.
- 3.2. The powers of the Association are further limited as set out in Schedule 1 to the Act, annexed hereto as Annexure 4.

4. MAIN BUSINESS AND OBJECTS

- 4.1. The main business of the Association is
 - 4.1.1. to maintain the common property and facilities;
 - 4.1.2. to provide and maintain services (including streets, water, sewerage, storm water and those parts of the electrical reticulation networks not owned by the electricity supply utility) which serve the Erven and/or units situated in Monaghan Farm;
 - 4.1.3. to control the aesthetic appearance of improvements within Monaghan Farm;
 - 4.1.4. to provide and make payment for measures to ensure the security, accessibility, and safety of Monaghan Farm, including its facilities and all persons in Monaghan Farm:
 - 4.1.5. to operate and manage the common facilities; and
 - 4.1.6. any and all activities ancillary and/or necessary in respect of the above business.
- 4.2. The main object(s) of the Association is to promote the group interests of the Members of the Association, by
 - 4.2.1. maintaining the common property and facilities;
 - 4.2.2. controlling the aesthetic appearance of improvements within Monaghan Farm;
 - 4.2.3. making, amending, repealing and/or enforcing rules and regulations for Monaghan Farm:
 - 4.2.4. preserving and enhancing the security, aesthetics, common property and the environment for the common good of Monaghan Farm;
 - 4.2.5. protecting the financial interests of the Members in the Monaghan Farm;

- 4.2.6. providing and maintaining all services (including streets, water, sewerage, storm water and electrical reticulation networks) which serve the Erven and units within Monaghan Farm; and
- 4.2.7. providing and paying for measures to ensure the security, accessibility and safety of Monaghan Farm and all persons therein.

5. OPTIONAL PROVISIONS OF THE COMPANIES ACT

- 5.1. The Company does not elect, in terms of section 34 (2), to comply voluntarily with the provisions of Chapter 3 of the Act.
- 5.2. The Company elects, in terms of section 30(2)(b)(ii), that the financial setatements in respect of any financial year may be either audited voluntarily or independently reviewed as may be required from time-to-time by ordinary resolution of the Members or the Board.

6. SPECIAL CONDITIONS

The special conditions which apply to the Association are –

- 6.1. the income and property of the Association whencesoever derived shall be applied solely towards the promotion of its main business and objects and no portion thereof shall be paid or transferred, directly or indirectly, by way of dividend, bonus or otherwise howsoever to the Members of the Association or to a holding company or a subsidiary, provided that nothing herein contained shall prevent the payment in good faith of reasonable remuneration to any officer or servant of the Association or to any Member thereof in return for any services actually rendered to the Association;
- 6.2. The Association shall be wound-up, deregistered or dissolved, subject to Item 1 (4) (a) and (b) of Schedule 1 to the Companies Act, subject further thereto that:
 - 6.2.1. The Members, if any, or the Association's directors if there are no Members, shall by way of special resolutions at or immediately before the time of dissolution; or
 - 6.2.2. Failing which the Court shall determine one or more non-profit companies, registered external non-profit companies carrying on activities within the Republic, voluntary associations or non-profit trusts to which the net assets of the Association shall be distributed.
- 6.3. Distributions referred to in Article 6.2 above shall only be made to non-profit companies, registered external non-profit companies carrying on activities within the Republic, voluntary associations or non-profit trusts who have been exempt from income tax in terms of section 10 (1) (e) (i) (cc) of the Income Tax Act.

- 6.4. The Association is not permitted to distribute its funds to any person other than to a similar association of persons.
- 6.5. Funds available for investment may only be invested with a financial institution as defined in section 1 of the Financial Services Board Act 97 of 1990 and in securities listed on a stock exchange as defined in section 1 of the Stock Exchanges Control Act 1 of 1985.

7. **RESOLUTIONS**

- 7.1. Decision by resolution of the Members or Board, as applicable, shall be effected by
 - 7.1.1. ordinary resolution being a resolution requiring that those voting in favour represent a simple majority of those who are present, in person or by proxy, and who are entitled to vote; or
 - 7.1.2. special resolution being a resolution requiring that those voting in favour represent not less than 75% (seventy five percent) of those who are present, in person or by proxy, and entitled to vote; or
 - 7.1.3. unanimous resolution being a resolution requiring that all of the Members (i.e. 100%) vote in favour of the resolution; or
- 7.2. Resolution may also be effected by Round Robin Resolution. "Round Robin Resolution" means a resolution passed, or which is to be passed, other than at a Board or Members' Meeting, as applicable, and which was or is voted favourably upon by all those entitled to vote by signing a written resolution in counterparts.

8. FINANCIAL YEAR

The financial year end of the Association is the last day of February of each year.

9. ACCOUNTING RECORDS AND FINANCIAL STATEMENTS

- 9.1. Notwithstanding the provisions of Article 5 above, the Association shall keep all such accurate and complete accounting records, in English, as are necessary to enable the Association to satisfy its obligations in terms of
 - 9.1.1. the Act;
 - 9.1.2. the Regulations;
 - 9.1.3. any other law with respect to the preparation of financial statements to which the Association may be subject; and

- 9.1.4. this MOI.
- 9.2. The Association shall each year prepare annual financial statements within 6 (six) months after the end of its financial year, or such shorter period as may be appropriate to provide the required notice of an Annual General Meeting.
- 9.3. The Association shall each year appoint an auditor at its Annual General Meeting. If the Association appoints a firm as its auditor, any change in the composition of the members of that firm shall not by itself create a vacancy in the office of auditor.
- 9.4. In every financial year the financial statements of the Association shall be audited.
- 9.5. The duties of the auditors shall be regulated in accordance with Part C of Chapter 3 of the Act, taking into account Regulations 26 to 29 of the Regulations, provided that the auditors shall be obliged to endeavour to finalise their duties within 3 (three) months after the end of the Association's financial year.
- 9.6. At each Annual General Meeting, the Directors shall lay before the Association the financial statements for the previous financial year, consisting of income statements, cash flows and balance sheets for such previous financial year as well as the financial year immediately preceding such previous financial year, of the Association. The balance sheet for the immediately preceding financial year shall be accompanied by the reports of the Directors and the auditors.
- 9.7. The annual financial statements shall be prepared on a basis that is not inconsistent with any unalterable provision of the Act, and shall satisfy, as to form and content, the financial reporting standards as determined by the Act.
- 9.8. The Association shall comply with its relevant obligations with regards to the auditing of the financial statements.

10. MEMBERSHIP AND CLASSES OF MEMBERS

- 10.1. The Association shall have a single class of members with voting rights.
- 10.2. Membership of the Association shall be limited to -
 - 10.2.1. any person who is, or persons who are, in terms of the Deeds Registries Act, reflected in the records of the relevant Deeds Office concerned as the registered owner or owners of land within Monaghan Farm, and

- 10.2.2. the Rhenosterspruit Conservancy who shall have be deemed to be one Member and have the corresponding voting rights of one fully paid-up Member of the Association.
- 10.2.3. the operators of the school on Erf 289, currently Curro Holdings, who are contractually bound to contribute the equivalent of 12 (twelve) residential levies and who, if fully paid-up, shall be deemed to be 12 (twelve) Members and have the corresponding voting rights in the Association.
- 10.3. A person who is, or persons who are, reflected in the records of the Deeds Office as the registered owner, or owners, of land within Monaghan Farm, shall, subject to proof to the contrary, be deemed to be the lawful registered owner or joint owners, as the case may be, thereof.
- 10.4. Where land within Monaghan Farm is owned by more than one person, all the registered owners shall together be deemed to be one Member of the Association and have the rights and obligations of one Member of the Association.
- 10.5. Concurrently with every offer to purchase land within Monaghan Farm, every prospective purchaser must also consent in the prescribed form to his membership of the Association and in which prescribed form such Member shall also choose a *domicilium citandi et executandi* address (address for service). Thereafter, and upon registration of the land into the name of such prospective purchaser, he shall *ipso facto* become a Member of the Association, and when he ceases to be the owner of land within the Monaghan Farm, he shall *ipso facto* cease to be a Member of the Association.

10.6. In the event of -

- 10.6.1. land being owned by more that one person, the co-owners concerned shall elect one of their number as liaison person for the land concerned, and shall notify the Association of the name and address of such liaison person;
- 10.6.2. a Member being a legal entity or a trust, such entity shall nominate a natural person to act as its representative and shall advise the Association of such nomination from time to time. The natural person so nominated shall, in a personal capacity, be liable jointly and severally for the obligations of the Member in terms of this MOI.
- 10.7. No Member shall part with occupation of his land or any improvements thereon, whether temporarily or otherwise, unless he has agreed with the proposed occupier of the improvement thereon, as a stipulatio alteri (a contact for the benefit of a third party) in favour

- of the Association, that such occupier shall be bound by all the terms and conditions of this MOI.
- 10.8. No Member may rezone land or in any way change the use for which land may be utilised whether by way of rezoning, change of land use rights or a consent use or otherwise, irrespective of whether such consent is granted by any lawful authority, without approval of the Association.
- 10.9. No Member may subdivide or consolidate land, nor erect a second dwelling on land, irrespective of whether any lawful authority grants permission for such subdivision or consolidation or second dwelling, without approval by the Association.
- 10.10. A registered owner of land may not resign as a Member of the Association;
- 10.11. The rights and obligations of a Member shall not be transferable and every Member shall:
 - 10.11.1. further, to the best of his ability, the objects and interests of the Association;
 - 10.11.2. observe all rules made by the Association or the Board;
 - provided that nothing contained in this MOI shall prevent a Member from ceding his rights in terms of this MOI as security to the mortgagee of that Member's land.
- 10.12. A failure by a Member to observe the provisions of this MOI and all rules made by the Association or the Board, shall render such Member liable to a fine as stipulated in the Rules and determined by the Board from time to time.
- 10.13. The Association shall be entitled to bind the Members to contribute by way of subscriptions and levies towards the funds of the Association, and to enforce payment of, and to collect and receive from Members, such contributions and levies.
- 10.14. No Member shall be entitled to any of the privileges of membership of the Association, including the right to vote at a Members' meeting, unless and until he shall have paid every levy contribution together with interest thereon which may be due and payable to the Association from whatsoever cause arising.
- 10.15. The Association shall keep a register of Members which shall include, as a minimum
 - 10.15.1. The full names of the Member, or Members if the land is owned by more than one person;

- 10.15.2. The identity number, or registration number, of the Member, or Members, or, if the Member is, or the Members are, a foreign national, or if the Members are foreign nationals, such Member's, or Members', passport number;
- 10.15.3. All land registered in the name of the Member, or Members,;
- 10.15.4. The chosen *domicilium citandi et executandi* (address for service) of such Member, or Members, for purposes of service of any and all legal process; and
- 10.15.5. An e-mail address, postal address, physical address and/or a fax number, for purposes of forwarding any notice, document or statement, capable of being transmitted electronically, in terms of this MOI or the Act.
- 10.16. It shall be the Members' obligation to ensure that the information in the register of Members is accurate and true. Should any change occur with regards to that information, the Member, or Members, shall duly communicate such change to the Association, in writing, within 14 (fourteen) days from date of change. Should a Member or Members fail to update their information, then any service at the registered chosen domicilium citandi et executandi (address for service) or delivery of notices or documents to the addresses referred to in Article 10.15.5 above, shall be valid.
- 10.17. The Association shall bear no responsibility for any inaccurate information provided to it by its Members.
- 10.18. The Developer shall cease to be a Member of the Association when it is no longer the registered owner of land within Monaghan Farm.

11. LEVIES AND CONTRIBUTIONS

- 11.1. The Board shall from time to time impose levy contributions, upon Members for the purpose of meeting all the expenses which the Association has incurred or to which the Board reasonably anticipates the Association will be liable or will incur (including expenditure of a capital nature) in the pursuance of the main business and objects of the Association and whether by way of
 - 11.1.1. maintenance, repair, improvement and keeping in good order and condition common facilities, streets, storm water system and public lighting, reception buildings and parking areas, and/or any other common property and/or facilities;
 - 11.1.2. for the payment of all rates and other charges payable by the Association to the local authority in respect of the common property and/or for the services rendered to the Association by the local authority;

- 11.1.3. for payment of the salaries and/or wages of the employees of the Association; and
- 11.1.4. generally for the payments of all expenses necessarily or reasonably incurred in connection with the management of the Association and its affairs.
- 11.2. In pursuance of the main business and objects of the Association, the Board
 - 11.2.1. shall prepare an itemised estimate of income and expenditure (the "annual budget") in respect of the ensuing financial year, prior to the commencement of such ensuing financial year; and
 - 11.2.2. shall include in the annual budget an amount to be transferred to and held in reserve ("the reserve fund") to meet anticipated future expenditures not of an annual nature, including reasonable provision for the future maintenance, repairs and replacements of capital equipment and the repair of roads and storm-water facilities.
- 11.3. The Board shall prepare the annual budget for each financial year, and
 - 11.3.1. where application of that budget requires an increase to Members' prevailing levy contributions of not greater than 50% (fifty percent) above CPI, may adopt that budget and impose any concomitant increase to the levy contribution, or
 - 11.3.2. where application of that budget requires an increase to the Members' prevailing levy contribution of greater than 50% (fifty percent) above CPI, shall present that budget together with the proposed increase to the levy contribution for approval, or otherwise, by ordinary resolution of the Member's at the Annual General Meeting.
- 11.4. The Board shall, whenever it shall consider it appropriate, be entitled to recover all levy contributions from Members from time to time and at the end of each financial year, or as soon thereafter as reasonably possible, notify each Member at the address chosen by him, of the nature and amount of the relevant levy contributions paid and required to be paid by Members and required by the Association to meet the expenses during the following financial year.
- 11.5. The Board may in its absolute discretion differentiate between one type of scheme to another and may similarly determine that the portion of the total expenses in terms of the annual budget be apportioned in different ratios in respect of differing schemes or units utilised for differing schemes, depending on the nature of the intended or actual use thereof.
- 11.6. Every levy contribution, shall be payable in such manner as the Board may from time to time determine.

- 11.7. In the event of the Board, for any reason whatsoever, failing to prepare and serve notice as envisaged in the above Articles 11.4 timeously, every Member shall until receipt of such notice continue to pay the relevant levy contributions previously imposed, and shall after such notice immediately pay such adjusted levy contributions as may be specified in the notice.
- 11.8. The Board may from time to time propose that special contributions be levied upon all Members in respect of the Association's capital or operating expenses and which special levies shall be imposed only by ordinary resolution of the Members.
- 11.9. The Board may from time to time levy additional contributions upon specific Members for services provided (for example, but not limited to, the removal of building rubble and the cleaning of land or for non-compliance with the MOI or Rules).
- 11.10. The Board shall, in addition to such other rights as the Association may have in law as against its Members, determine the rate of interest from time to time chargeable upon levy contributions in arrears provided that such rate of interest shall not exceed the rates laid down in terms of the National Credit Act 34 of 2005, as amended, as if the arrear levy contribution were a money lending transaction as defined in the said Act.
- 11.11. Any amount due by a Member by way of levy contributions, together with any interest owed by the Member to the Association, shall be a debt due by him to the Association. The obligation of a Member to pay all levy contributions and interest shall cease upon his ceasing to be a Member without prejudice to the Association's rights to recover all arrear levy contributions and interest. No levy contributions or interest paid by a Member shall under any circumstances be repayable by the Association upon his ceasing to be a Member unless so determined by the Board in its sole and absolute discretion. A Member's successors in title shall be liable as from the date upon which he becomes a Member of the Association, to pay the levy contribution and interest thereon attributable to the interest to which he succeeds.
- 11.12. In calculating the contributions payable by any Member, the Board shall as far as reasonably practical
 - 11.12.1. assign costs arising directly out of land itself to the Member owning such land or interest thereon;
 - 11.12.2. assign a proportion of costs attributable generally to a particular scheme to the owners of units in that scheme pro-rata to the participation quota of the Member concerned; and

- 11.12.3. assign those costs for services attributable to Monaghan Farm as a whole, generally to the owners of land equally.
- 11.13. The Directors shall be empowered to impose penalty levies on owners who have failed to comply with the obligation to build as prescribed in the Rules of the Association, or, as the case may be, in the Deed of Sale by virtue of which the land was obtained which levies shall be payable in addition to levies normally applicable to the land.
- 11.14. The Developer shall, during the Development Period, make the following contribution to the funds of the Association:
 - 11.14.1. for the period 1 March 2014 to 31 July 2015, R350 000.00 (three hundred and fifty thousand Rand) per month; and
 - 11.14.2. thereafter; an amount equal to the sum of the number of proclaimed residential erven owned by the Developer multiplied by the prevailing normal monthly levy together with any special levy raised from time to time and applicable to each residential erven.

subject thereto that the Developer's contribution shall, during the development period, never be less than R50 000.00 (fifty thousand rand) per month.

- 11.14.3. Notwithstanding paragraphs 11.14, 11.14.1. and 11.14.2. above, from the date of proclamation of the Proposed Townships Monaghan Extension 8 to Monaghan Extension 17, the Developer of the Proposed Townships of Monaghan Extension 8 to Monaghan Extension 17, shall be exempted to contribute any monthly levy in terms of paragraph 11.1, or contribute any special levy in terms of paragraph 11.8, or contribute any additional contributions as per paragraph 11.9 in respect of each of the Erven owned by the Developer or each of the Erven capable of registration in the Deeds Office in the proposed Townships of Monaghan Extension 8 to Monaghan Extension 17.
- 11.14.4 Subsequent to the date of proclamation of the Proposed Townships Monaghan Extension 8 to Monaghan Extension 17, any cession or transfer of development rights and obligations from the Developer to another Developer post proclamation in respect of any Erven in the Proposed Townships Monaghan Extension 8 to Monaghan Extension 17, the new Developer will not to be entitled to be exempted to contribute any monthly levy in terms of paragraph 11.1, or contribute any special levy in terms of paragraph 11.8, or contribute any additional contributions as per paragraph 11.9 in respect of each of the Erven owned by the new Developer or any of the Erven held by the new Developer and capable of registration in the Deeds Office in the proposed Townships of Monaghan Extension 8 to Monaghan Extension 17.

- Subsequent to the date of proclamation of the Proposed Townships Monaghan Extension 8 to Monaghan Extension 17, and in the event of any development of any of the Erven by the Developer, other than for the installation of bulk services, in the Proposed Monaghan Extension 8 to Extension 17, in that event the Company waive the exemption of levy contributions as set out in paragraph 11.14.3 and the Developer will be liable in respect of the contribution of any monthly levy in terms of paragraph 11.1, or the contribution of any special levy in terms of paragraph 11.8, or the contribution any additional contributions as per paragraph 11.9 in respect of each of the Erven developed and owned by the Developer or each of the developed Erven capable of registration in the Deeds Office in the proposed Townships of Monaghan Extension 8 to Monaghan Extension 17.
- 11.14.6 Upon the sale and transfer of any Erven by the Developer within the proposed Townships of Monaghan Extensions 8 to Monaghan Extension 17 to any person who is, or persons who are, in terms of the Deeds Registries Act, reflected in the records of the relevant Deeds Office concerned as the registered owner or owners of an Erf within the Townships of Monaghan Extensions 8 to Monaghan Extension 17, will make the registered owner or owners, subject to paragraph 11.14.7, liable to contribute any monthly levy in terms of paragraph 11.1, or contribute any special levy in terms of paragraph 11.8, or contribute any additional contributions as per paragraph 11.9 in respect of each of the Erven owned by that person from date of transfer.
- 11.14.7 The liability to contribute levies of any person who is, or persons who are, in terms of the Deeds Registries Act, reflected in the records of the relevant Deeds Office concerned as the registered owner or owners of any Erf within Monaghan Extensions 8, 9, 10,11, 12 and 13, will be 75% (seventy-five per centum) of the levy contributions imposed by the Board in terms of paragraph 11.1 on the remaining members of the Company for a period of (5) five years from date the special resolution carries.
- 11.14.8 Any person who is, or persons who are, in terms of the Deeds Registries Act, reflected in the records of the relevant Deeds Office concerned as the registered owner or owners of any Erf within Monaghan Extensions 8 to Monaghan Extensions 17 will be liable for a full contribution in respect of any special levy contributions imposed by the Board in terms of paragraph 11.8.
- 11.14.9 No voting rights shall attach to the proclaimed Erven of Proposed Townships of Monaghan Extension 8 to Extension 17 until such time as levies and special levies are payable by any person who is, or persons who are, in terms of the Deeds Registries Act, reflected in the records of the relevant Deeds Office concerned as the registered owner or owners of any Erf within Monaghan Extension 8 to Extension 17.

- 11.15. From such time as Developer proceeds with development of the future erven whether within the development period or not, Developer shall make contribution to the Association in:
 - 11.15.1. a monthly amount calculated as R20,000 escalated from 1 March 2015 at CPI to the 1st day of the month during which any works on the future erven or their related infrastructure is undertaken escalated further by CPI on each anniversary; plus
 - 11.15.2. any and all costs incurred each month by the Association arising from the works associated with the development of the future erven.
- 11.16. As from the date of proclamation of such future erven the Developer shall contribute an amount equal to the sum of the number of proclaimed future erven owned by the Developer multiplied by the prevailing normal monthly levy together with any special levy raised from time to time and applicable to each residential erven as calculated from the first day of the month within which proclamation of each of the erven was made; provided that should the Developer develop the future erven in phases, its obligation as set out in 11.15 shall remain in respect of any unproclaimed future erven.
- 11.17. In the event that the Developer elects to exercise its commercial rights under the approved town planning scheme whether within the development period or not, the Developer shall make a contribution to the Association in terms of 11.15.1 and 11.15.2 above together with a levy equal to one residential levy contribution for each 350 m2 (three hundred and fifty square meters) of floor area, or part thereof, contained in each commercial development.
- 11.18. The levies in respect of units within a sectional title scheme shall not exceed 50 % (fifty percent) of the normal levy determined by the Board.
- 11.19. Levy stabilisation fund and re-sale fee
 - 11.19.1. The Board shall establish a Levy Stabilisation Fund. The Levy Stabilisation Fund is to ensure that future levies remain within acceptable levels and that the need for raising any special levies by the Board is reduced.
 - 11.19.2. The Levy Stabilisation Fund shall fall outside of the other levy provisions provided for elsewhere in this Memorandum of Incorporation and the contributions to the Levy Stabilisation Fund shall be funded from the re-sale fee raised in terms of this Article and paid into a separate banking account which the Association shall open and keep with a financial institution.
 - 11.19.3. Any amount standing to the credit of the Levy Stabilisation Fund may only be used to fund expenditure of a capital nature and / or any extraordinary expenditure that may be incurred by the Association in carrying out its business and objects under the provisions of this Memorandum of Incorporation.
 - 11.19.4. It shall be a condition of membership of the Association and for the approval of the re-sale of an erf or unit or interest therein or of a controlling or beneficial interest in

- a juristic person or a trust that Members are liable to pay a contribution to the Levy Stabilisation Fund as determined herein below.
- 11.19.5. The contribution to the Levy Stabilisation Fund is hereinafter referred to as the *resale fee*.
- 11.19.6. The liability of Members to make payment of the re-sale fee is made payable in the event of an erf or unit or interest therein or of a controlling or beneficial interest in a juristic person or a trust being sold. In clarification of the latter instance, the resale fee shall be payable in the event of a Member being a juristic person or a trust and a controlling or beneficial interest in such juristic person or trust being sold to a new holder thereof, notwithstanding that the juristic person or the trust has remained the owner of the erf or unit or interest therein.
- 11.19.7. The re-sale fee shall be a once-off payment in respect of the erf, unit, interest therein, controlling, or beneficial interest and shall be determined as follows:
 - 1% (one percent) of the sale price plus VAT on the determined percentage amount; or
 - 1% (one percent) of the value plus VAT on the determined percentage amount, in the event of a transaction where no sale price is specified. The sale price shall be the sale price as is evidenced from the agreement of sale, power of attorney to pass transfer or any other relevant document as may be requested by the Association.

The value shall be the value as is evidenced from any relevant document as may be requested by the Association.

Where the sale price or value is subject to VAT, such VAT shall be excluded, and the sale price or value shall be the amount net of VAT.

No deduction or set-off of any amounts whatsoever are allowed to be made in determining the re-sale fee.

- 11.19.8. The Association shall have the power to determine the re-sale fee and the same shall be payable in circumstances where a transaction, operation or scheme has been entered into with the purpose of circumventing the payment of the re-sale fee. This may include for example, but not be limited to, donations, exchanges or similar disposals.
- 11.19.9. The Association shall not approve the transfer of an erf or unit or interest therein or the transfer of a controlling or beneficial interest in a juristic person or a trust, unless the re-sale fee has been paid or the payment of the re-sale fee has been secured to the satisfaction of the Board of the Association.
- 11.19.10. The re-sale fee shall be due, owing and payable on or before the registration of transfer of an erf or unit or interest therein or on or before the transfer of a controlling or beneficial interest in a juristic person or a trust, unless otherwise agreed to in writing by the Board of the Association.
- 11.19.11. No re-sale fee shall be payable in the event of a transfer of ownership arising from testate or intestate succession or a divorce court order (provided that the transfer is from one spouse to another). In the event of a court ordered execution sale of a unit or a sale of a unit following the liquidation or the sequestration of a member, the Board is empowered to resolve within its sole and absolute discretion on the exemption of the re-sale fee payable. The Board of the Association shall be entitled to determine such further exemptions in its sole and absolute discretion.

- 11.19.12. Members or any other persons shall not be entitled to a refund of the re-sale fee or of any portion thereof.
- 11.19.13. Should the re-sale fee require adjustment, the Board of the Association shall present the proposed adjustment for approval or otherwise by special resolution to the Members at an Annual General Meeting.
- 11.19.14. The provisions of this Memorandum of Incorporation, including Articles 11.10, 11.11 and 16 thereof shall, mutatis mutandis, apply to the re-sale fee.
- 11.19.15. The provisions of this Article 11.19 shall in no way serve to exclude the payment of any amounts as are provided for in Article 11 of this MOI.
- 11.19.16. The provisions of this Article 11.19 shall prevail in the event of a conflict with any other provisions contained elsewhere in this MOI, the Rules, the Regulations of the Association or any other policy and procedure documents of the Association which deal with the same subject matter.
- 11.19.17. A Member undertakes to notify a purchaser of an erf or unit or interest therein or of a controlling or beneficial interest in a juristic person or a trust of the provisions of this Article 11.19. and that the same is a condition of membership of the Association."

12. AMENDMENTS TO THE MOI

- 12.1. In order to fulfil the requirements for exemption from income tax on levies payable to, or accruing to, the Association, any amendment to this MOI and must be submitted to the Commissioner of the South African Revenue Services.
- 12.2. The Board is, by ordinary resolution, -
 - 12.2.1. empowered to correct errors substantiated as such from objective evidence or which are self evident errors (including, but without limitation, *ejusdem generis*, spelling, punctuation, reference, grammar or similar defects) in the MOI, and
 - 12.2.2. amend this MOI in compliance with a court order

and shall publish a copy of any such correction effected by the Board on the Association's web-site.

- 12.3. This MOI may be amended at any time if a special resolution to amend it -
 - 12.3.1. is proposed by
 - 12.3.1.1. The Board: or

- 12.3.1.2. Members entitled to exercise at least 10% of the voting rights that may be exercised on such a resolution; and
- 12.3.2. is adopted at a Members' Meeting.
- 12.4. Within 10 (ten) business days after an amendment to the MOI has been effected as the Association must file a Notice of Amendment in the form required by Regulation 15, together with all relevant documents as contemplated in Section 16(7) of the Act together with the prescribed fee.

13. THE PROCEDURE FOR MAKING OF RULES

- 13.1. A rule issued in terms of section 15(3) of the Act
 - 13.1.1. must be consistent with the Act and this MOI, and any such rule that is inconsistent with the Act or this MOI is void to the extent of the inconsistency;
 - 13.1.2. takes effect on a date that is the later of-
 - 13.1.2.1. 10 (ten) business days after the rule is filed with the Commission; or
 - 13.1.2.2. the date, if any, specified in the rule; and
 - 13.1.3. is binding
 - 13.1.3.1. on an interim basis from the time it takes effect until it is put to a vote at the next Members' Meeting of the Association; and
 - 13.1.3.2. on a permanent basis only if it has been ratified by an special resolution of the Members.
- 13.2. If a rule that has been filed with the Commission is subsequently
 - 13.2.1. ratified by the Members, the Association must file a notice of ratification within 5 (five) business days in the prescribed manner and form; or
 - 13.2.2. not ratified when put to a vote
 - 13.2.2.1. the Association must file a notice of non-ratification within 5 (five) business days after the vote, in the prescribed manner and form; and
 - 13.2.2.2. the Board may not make a substantially similar rule within the ensuing 12 (twelve) months, unless it has been approved in advance by special resolution of the Members.

- 13.3. Any failure to ratify a rule of the Association does not affect the validity of anything done in terms of that rule during the period that it had an interim effect.
- 13.4. Where an amendment to the Rules is one to correct self evident errors (including, but without limitation *ejusdem generis*, spelling, punctuation, reference, grammar or similar defects), the Board shall publish a copy of any such correction effected by the Board on the Association's website..
- 13.5. The Rules of the Association in existence at the time that this MOI is registered at the Commission, shall be binding as if such rules were made in accordance with this MOI and in terms of the Act

14. REGULATIONS, & HOUSE RULES

- 14.1. The Board may from time to time make, amend or repeal any necessary or incidental regulations relating to the Association, and the authority of the Board in this regard is limited or restricted only in that the Members, by special resolution, may, at a Members' Meeting specifically called for such purpose or in respect of which notice has been given, require that certain regulations be made, amended or repealed by the Board, and/or that any regulations made or amended be entrenched. Any regulation so entrenched shall only be capable of being amended or repealed by the Members of the Association by special resolution.
- 14.2. Subject to any restriction imposed or direction given at a Members' meeting of the Association, the Board may from time to time make, amend, or repeal house rules to govern matters including
 - 14.2.1. the means of access to, and the rules by which Members and member of their households, guests or contractors, may gain access, to Monaghan Farm, including the registration of guests and visitors;
 - 14.2.2. the limitation of the number of persons and vehicles able to visit a Member;
 - 14.2.3. the use by Members and their households, guests and lessees of any roads, driveways, water areas and open spaces including the right to prohibit, restrict or control the use of any of the roads, driveways and open spaces on the common property or any portions thereof as may from time to time be necessary or expedient;
 - 14.2.4. the use of parking areas where ever situated;
 - 14.2.5. the use of land;

- 14.2.6. the placing or fixing of movable objects, including ornamentation or embellishment upon the exterior of a building, including the power to remove any such objects;
- 14.2.7. the preservation of the environment including the right to control vegetation and the right to prohibit and/or control the erection of fences, whether upon or within the boundaries of any part of Monaghan Farm and/or individual land;
- 14.2.8. the right to prohibit, restrict or control the keeping of pets in and upon Monaghan Farm and any animal regarded as dangerous or a nuisance;
- 14.2.9. the conduct of any person within Monaghan Farm for the prevention of nuisances of any nature including disturbing the peace;
- 14.2.10. the use of all common properties, water areas, entertainment and recreation areas, amenities and facilities including the right to make a reasonable charge for the use thereof;
- 14.2.11. the furtherance and promotion of the objective of the Association and/or for the better management of the affairs of the Association and/or for the protection and advancement of the interests of Members and/or residents in Monaghan Farm;
- 14.2.12. standards and guidelines for architectural design of all buildings, outbuildings, structures of any nature, swimming pools, tennis courts and all additions and alterations to any such buildings, outbuildings or structures erected or to be erected in Monaghan Farm, and in particular to control the design of the exterior of such buildings, outbuildings or structures and the materials used on such exteriors to ensure an attractive, aesthetically pleasing character to all the buildings in Monaghan Farm;
- 14.2.13. the positioning within any land of all buildings, outbuildings, structures of any nature, swimming pools, tennis courts and all additions and alterations thereof;
- 14.2.14. the standards and guidelines for the design of all site works, buildings, structures, installations and projections on the properties in Monaghan Farm, including aerials, air conditioners, cellphone receptors, pergolas, side- walks, swimming pools, tennis courts, awnings, jacuzzi's, carports, paved pathways and landscaping features and works as well as the parking of vehicles; and
- 14.2.15. the access to, or use of, any other services not specifically provided in this Article.

15. PUBLICATION OF RULES, REGULATIONS, & HOUSE RULES

- 15.1. Rules, Regulations, & House Rules and amendments thereto shall
 - 15.1.1. be published on the Association's website and a notice of availability of that document, record or statement, summarising its content and satisfying any prescribed requirements, in accordance with Regulation 6 of the Regulations to the Act, shall be delivered to each intended recipient of the document, record or statement, together with instructions for receiving the complete document, record or statement; and
 - 15.1.2. the notice of availability shall be addressed to the respective Member at his address shown in the register of Members; and
 - 15.1.3. be deemed to have been received by the Member to whom it is addressed at the time of delivery thereof, or on the 7th day following the posting thereof (excluding the day upon which it was posted) in the Republic, or in the case of a fax, on the date and time stipulated on the fax transmission report thereof, or in the case of an e-mail, on the date and time recorded by the computer of the Association.

16. ENFORCEMENT & DISPUTE

- 16.1. For the enforcement of any of the existing or future rules or regulations the Board may
 - 16.1.1. give notice to any Member requiring him to remedy any such breach within a reasonable period as the Board may determine;
 - 16.1.2. take or cause to be taken such steps as it may consider necessary to remedy any breach of which the Member may be guilty and debit the costs of so doing to the Member concerned:
 - 16.1.3. impose a system of fines or penalties, as the case may be, the amounts of which fines and penalties shall be stipulated in the Regulations, to be determined by the Board; or
 - 16.1.4. take such action including proceedings in court as it may deem necessary.
- 16.2. In the event of the Board taking any legal steps or instituting any legal proceedings against any Member or resident within Monaghan Farm for the enforcement of this MOI, any of the rules and/or regulations or rights of the Association in terms of this MOI, including the collection of any arrear levies, interest and/or any other debt due by the Members to the Association, the Association shall be entitled to appoint such attorneys and counsel as it deems necessary and to recover all legal costs so incurred from the Member or resident

- concerned, calculated as between attorney and own client, and to debit such Member's account with such legal costs accordingly if necessary.
- 16.3. In the event of any breach of the rules by any Member's household or his guests, contractors or lessees, the Board may take or cause to be taken such steps against the Member personally, or against the person actually committing the breach, as the Board may in its sole discretion deem necessary.
- 16.4. In the event of any Member disputing the fact that he has committed a breach of any provision of this MOI, or a rule or regulation of the Association, a committee of three persons appointed by the Board for the purpose shall adjudicate upon the issue at such time and in such manner and according to such procedure as the Chairman may direct: Provided that natural justice shall be observed. A person who is not satisfied with the decision of the said committee may appeal to the Board whose decision is final.
- 16.5. Any fine imposed upon a Member in terms of these Articles shall be deemed to be a debt by the Member to the Association and, if not paid, shall be recoverable by ordinary civil process.

17. MEMBERS' MEETINGS

- 17.1. The Association elects to hold the following meetings additional to those required by the Act:
 - 17.1.1. The Association shall, within 6 (six) months after the end of each financial year, hold a general meeting of Members, to be referred to as the Annual General Meeting, in addition to any other Members' meetings during that year and shall specify the meeting as such in notices to Members calling such meeting.
- 17.2. An Annual General Meeting, must, at a minimum, provide for the following business to be transacted:
 - 17.2.1. Presentation and consideration of
 - 17.2.1.1. the directors' report;
 - 17.2.1.2. the audited financial statements for the immediately preceding financial year; and
 - 17.2.1.3. the auditors' report.
 - 17.2.2. Election of Directors, to the extent required by the Act or this MOI;

- 17.2.3. The noting of contributions and/or levies for the ensuing financial year; and
- 17.2.4. Appointment of an auditor for the ensuing financial year; and
- 17.2.5. Any matters raised by the Members upon at least 7 (seven) days' written notice to the Association.
- 17.3. The Board may in their discretion decide whether a Members' meeting may be conducted entirely by electronic communication or in person, and if in person, whether one or more Members, or proxies for Members, may participate by electronic communication in all or part of a Members' meeting that is held in person within the precincts of Monaghan Farm at such time and location as the Board may decide.
- 17.4. All Members' Meetings other than Annual General Meetings shall be called Special General Meetings.
- 17.5. The Board may whenever it deems fit convene a Special General Meeting and must call such a meeting, in terms of section 61(3) of the Act, if
 - 17.5.1. one or more written and signed demands for such a meeting are delivered to the Association; and
 - 17.5.2. each such demand describes the specific purpose for which the meeting is proposed; and
 - 17.5.3. in aggregate, demands for substantially the same purposes are made and signed by Members, as of the earliest time specified in any of those demands, of at least 5 % (five percent) of the voting rights entitled to be exercised in relation to the matter proposed to be considered at the meeting.
- 17.6. The Board may, as an alternative to calling a Special General Meeting, and insofar as is practical and in accordance with the provisions of the Act, -
 - 17.6.1. submit, in compliance with the provisions of section 60 of the Act, a Round Robin Resolution for consideration to the Members entitled to exercise voting rights in relation to the resolution; and
 - 17.6.2. such resolution may be voted on in writing by the Members entitled to exercise voting rights in relation to the resolution within 20 (twenty) days after the resolution was submitted to them.
- 17.7. The business required to be discussed during an Annual General Meeting may not be put to vote by Round Robin Resolution.

17.8. Clause deleted. (by way of Special Resolution at Annual general Meeting 28 August 2021).

18. NOTICE OF MEMBERS' MEETING

- 18.1. Notice of Member's meetings shall be in accordance with section 62 of the Companies Act.
- 18.2. A Members' meeting shall be called by at least 15 (fifteen) Business Days' notice delivered by the Association to all Members entitled to vote or otherwise entitled to receive notice.
- 18.3. The Association may call a Members' Meeting with less notice but such a Members' Meeting may proceed only if every person who is entitled to exercise voting rights in respect of any item on the meeting agenda
 - 18.3.1. is present at the Members' meeting; and
 - 18.3.2. votes to waive the required minimum notice of the Members' Meeting.
- 18.4. A Member entitled to vote who is present at a meeting, either in person or by proxy
 - 18.4.1. Is regarded as having received or waived notice of the meeting, if at least the required minimum notice was given; and
 - 18.4.2. has a right to --
 - 18.4.2.1. allege a material defect in the form of notice for a particular item on the agenda for the meeting; and
 - 18.4.2.2. participate in the determination whether to waive the requirements for notice, if at least the required minimum notice was given, or to ratify a defective notice; and
 - 18.4.3. except to the extent set out in Article 18.4.2 is regarded to have waived any right based on an actual or alleged material defect in the notice of the Members' Meeting.
- 18.5. A notice of a Members' Meeting must be in writing, in plain language and must include
 - 18.5.1. the date, time and place for the meeting, and the Record Date for the meeting;
 - 18.5.2. the general purpose of the Meeting, and any specific purpose contemplated in Article 17.5.2, if applicable;
 - 18.5.3. a copy of any proposed resolution of which the Association has received notice, and which is to be considered at the meeting, and a notice of the percentage of voting rights that will be required for that resolution to be adopted;

- 18.5.4. in the case of an Annual General Meeting of the Association
 - 18.5.4.1. the financial statements to be presented or a summarised form thereof; and
 - 18.5.4.2. directions for obtaining a copy of the complete annual financial statements for the preceding financial year; and
- 18.5.5. a reasonably prominent statement that -
 - 18.5.5.1. a Member entitled to attend and vote at the meeting is entitled to appoint a proxy to attend, participate in and vote at the meeting in the place of the Member;
 - 18.5.5.2. a proxy need not also be a Member of the Association; and
 - 18.5.5.3. Section 63 (1) requires that meeting participants provide satisfactory identification.
- 18.6. A meeting may proceed notwithstanding a material defect in the giving of the notice of such meeting, subject to Article 18.7, only if every person who is entitled to exercise voting rights in respect of any item on the meeting agenda is present at the meeting and votes to approve the ratification of the defective notice.
- 18.7. If a material defect in the form or manner of giving notice of a meeting relates only to one or more particular matters on the agenda for the meeting
 - 18.7.1. any such matter may be severed from the agenda, and the notice remains valid with respect to any remaining matters on the agenda; and
 - 18.7.2. the meeting may proceed to consider a severed matter, if the defective notice in respect of that matter has been ratified.
- 18.8. An immaterial defect in the form or manner of giving notice of a Members' Meeting, or an accidental or inadvertent failure in the delivery of the notice to any particular Member to whom it was addressed if the Association elects to do so, does not invalidate any action taken at the meeting.

19. CHAIR, QUORUM & VOTING

19.1. The Chairman of the Board shall preside as Chairman at every Members' Meeting. If there is no such Chairman, or if at any Members' Meeting s/he is not present within 15 (fifteen) minutes after the time appointed for holding the Members' Meeting or is unwilling to act as

Chairman, the members entitled to vote which are present shall select a Director present at the Members' Meeting, or if no Director be present at the Members' Meeting, or if all the Directors present decline to take the chair, the Members present and entitled to vote shall select one of their number which is present to be Chairman of the Members' Meeting.

- 19.2. A meeting may not begin, and a matter to be decided at the meeting may not begin to be considered, unless a quorum has been established such that sufficient persons are present at the meeting (in person or by proxy) to exercise, in aggregate, at least 20 % (twenty percent) of all of the voting rights that are entitled to be exercised in respect of at least one matter to be decided at the meeting.
- 19.3. If within 1 (one) hour after the appointed time for the meeting to begin,
 - 19.3.1. a quorum for that meeting to begin has not been established, the meeting is postponed without motion, vote or further notice, for 1 (one) week;
 - 19.3.2. a quorum for consideration of a particular matter has not been established
 - 19.3.2.1. if there is other business on the agenda of the meeting, consideration of that matter may be postponed to a later time in the meeting without motion or vote; or
 - 19.3.2.2. if there is no other business on the agenda of the meeting, the meeting is adjourned for 1 (one) week, without motion or vote.
- 19.4. A Members' Meeting may not be adjourned beyond the earlier of
 - 19.4.1. The date that is 45 (forty five) Business Days after the Record Date; or
 - 19.4.2. the date that is 30 (thirty) Business Days after the date on which the adjournment occurred.
- 19.5. No further notice is required to be given by the Association of a meeting that is postponed or adjourned for failure to establish a quorum unless the location for the meeting is different from
 - 19.5.1. the location of the postponed or adjourned meeting; or
 - 19.5.2. a location announced at the time of adjournment, in the case of an adjourned meeting.

- 19.6. After a quorum has been established for a meeting, or for a matter to be considered at a meeting, the meeting may not continue, or the matter may not be considered, if the members present, for whichever reason, no longer constitute a quorum.
- 19.7. At every Members' Meeting:
 - 19.7.1. every Member present in person or by proxy and entitled to vote shall have one vote for each ordinary and special resolution for each land, developed or not, registered in his name;
 - 19.7.2. co-owners having rights or interests in respect of the same land shall, unless otherwise determined by prior resolution of the Board, jointly have one vote for each ordinary and special resolution;
 - 19.7.3. the Developer shall, subject to clause 19.8, have one vote in respect of each residential Erf registered in its name and which number of votes shall, during the development period, be not less than 10 (ten).
- 19.8. Save as expressly provided for in this MOI and unless specifically permitted otherwise by the Chairman, no person other than a Member duly registered, and who shall have paid every contribution, levy and other sum, if any, which shall be due and payable to the Association in respect of or arising out of his membership, and who is not under suspension, shall be entitled to be present or to vote on any matter either personally or by proxy, at any Members' Meeting.
- 19.9. No objection shall be raised as to the admissibility of any vote except at the Members' Meeting or adjourned Members' Meeting at which the vote objected to is or may be given or tendered and every vote not disallowed at such Members' Meeting shall be valid for all purposes. Any such objection shall be referred to the Chairman of the Members' Meeting, whose decision shall be final and conclusive.
- 19.10. All voting shall be done by way of polling conducted forthwith and in such manner as the Chairman directs, and the result of the poll shall be deemed to be the resolution of the Members' Meeting. Scrutineers may be appointed by the Chairman to declare the result of the poll and if appointed, their decision, which shall be given by the Chairman of the Members' Meeting, shall be deemed to be the resolution of the Members' Meeting.
- 19.11. Every resolution and every amendment of a resolution proposed at a Members' Meeting shall be seconded at the meeting and if not so seconded shall be deemed not to have been proposed.

- 19.12. In the case of equality of votes for and against any resolution, the Chairman shall have a second or casting vote.
- 19.13. Unless any Member present in person or by proxy at a Members' Meeting shall before the closure of the meeting have objected to
 - 19.13.1. any declaration made by the Chairman of the meeting as to the result of any voting at the meeting, by poll; and/or
 - 19.13.2. the propriety or validity of the procedure at such a meeting;

such declaration by the Chairman shall be deemed to be a true and correct statement of the voting, and the meeting shall in all respects be deemed to have been properly and validly constituted and conducted, and an entry in the minutes to the effect that any motion has been carried or lost, with or without a record of the number of votes recorded in favour of or against such motion, shall be conclusive evidence of the vote so recorded if such entry confirms the declaration made by the Chairman of the meeting as to the result of any voting at the meeting. All resolutions by the Members must be dated and sequentially numbered in the minutes of the Association.

- 19.14. In addition to resolutions required by the Act to be approved by special resolution, the following resolutions and/or business shall require approval by the Members by way of a special resolution
 - 19.14.1. The use or change of use of the common facilities; and
 - 19.14.2. The application for rezoning of private open space as may be required for the provision of infrastructural or communications services.
- 19.15. The following shall require the approval of the Members by unanimous resolution -
 - 19.15.1. The application for rezoning of proclaimed private open space save as provided for in Article 19.14.2 above; and
 - 19.15.2. Any application for any increase, reduction, or subdivision of the 305 (three hundred and five) erven approved for Residential 1 development on Monaghan Farm.

20. PROXIES

20.1. A Member may be represented at a Members' Meeting by proxy, who need not be a Member of the Association. A proxy shall not be entitled to delegate his authority to act on behalf of the Member to another person. A Member may appoint a single person as proxy or two

persons concurrently as proxies, to act in the alternative should one of them not be able to be present at the meeting, provided that no person shall be entitled to accept and/or be the holder of, more than 5 (five) proxy appointments.

- 20.2. A proxy shall be appointed on a proxy form ("Proxy Form") prepared for that purpose by the Association and made available for use to appoint proxies.
- 20.3. The Proxy Form shall be signed by the Member concerned or his duly authorised agent, provided that where a Member is more than one person, a majority of those persons shall sign the Proxy Form on such Member's behalf. Where a Member is a company, the Proxy Form may be signed by the Chairman of the Board of Directors of that Company or by its Secretary, and where an association of persons, by the Secretary thereof, or in the case of a trust, a duly appointed Trustee thereof.
- 20.4. No proxy form shall be valid after the conclusion of the meeting at which the proxy was to be exercised or such later meeting held due to adjournment.
- 20.5. The form appointing a proxy and the power of attorney or other authority, if any, under which it is signed or a notarially certified copy of such power or authority shall be delivered to the Association 48 (forty eight) hours prior to the Members' Meeting, before the proxy exercises any rights of the Member entitled to vote at a Members' Meeting.
- 20.6. A vote given in accordance with the terms of an instrument of proxy shall be valid notwithstanding the death or mental disorder of the principal or revocation of the proxy or of the authority under which the proxy was executed, or the transfer of the Membership in respect of which the proxy is given, provided that no intimation in writing of such death, insanity, revocation or transfer as aforesaid shall have been received by the Association at its Registered Office before the commencement of the Members' Meeting or adjourned Members' Meeting at which the proxy is used.
- 20.7. If a proxy is received duly signed but with no indication as to how the Person named therein should vote on any issue, the proxy may vote or abstain from voting as he sees fit.

21. RECORD DATE

- 21.1. If the Board determines the Record Date, it may not be earlier than the date on which the Record Date is determined or more than 10 (ten) Business Days before the date on which the event or action, for which the Record Date is being set, is scheduled to occur.
- 21.2. If, at any time, the Board fails to determine a Record Date, the Record Date for the relevant matter is –

- 21.2.1. in the case of a Members' Meeting, the latest date by which the Association is required to deliver to Members entitled to vote, notice of that Members' Meeting; or
- 21.2.2. the date of the action or event, in any other case.
- 21.3. The Association must publish a notice of a Record Date for any matter by
 - 21.3.1. delivering a copy to each Member; and
 - 21.3.2. posting a conspicuous copy of the notice -
 - 21.3.2.1. at its registered office; or
 - 21.3.2.2. on the Association's web-site, if it has one.

22. DIRECTORS, ELECTION OF DIRECTORS AND CASUAL VACANCIES

- 22.1. There shall be a Board of Directors of the Association which shall (unless otherwise determined by special resolution in general meeting) consist of not less than 7 (seven) and not more than 10 (ten) Directors.
- 22.2. A Director shall be an individual but need not himself be a Member of the Association. A Director, by accepting his appointment to office as such, shall be deemed to have agreed to be bound by all the provisions of this MOI and the rules and regulations of the Association.
- 22.3. The Developer, during the Development Period, shall be entitled to appoint 2 (two) Directors ("Developer Directors"), which appointed Directors need not be elected by the Members, nor retire, and the Developer shall be entitled from time to time on notice to the Association, to terminate the appointment of any Director appointed by it in and to appoint such other person as a Director of the Association provided that the Developer shall indemnify the Association, its Directors and Members against any claims that a Director, appointed and subsequently removed by the Developer, may have.
- 22.4. Except in respect of a Developer Director, each of the Directors shall
 - 22.4.1. be elected at an Annual General Meeting and serve as a Director until the second anniversary of the Annual General Meeting at which he/she was elected following which he/she shall be deemed to have reitred from office but shall be eligible for re-election; or
 - 22.4.2. should a vacancy arise prior to an Annual General Meeting, a Director, on an interim basis, may

- 22.4.2.1. be elected by the members by Round Robin Resolution, or
- 22.4.2.2. be appointed by the Board,

save that any Director so elected or appointed shall automatically retire at the Annual General Meeting following such election or appointment but shall be eligible for re-election.

- 22.5. In addition to Article 22.4 above, at least one-third (1/3rd) of the Board, calculated without the inclusion of any Developer Director, shall retire immediately before each Annual General Meeting. Such retiring Directors shall be eligible for re-election.
- 22.6. There are no general qualifications prescribed by the Association for a Person to serve as a Director in addition to the requirements of the Companies Act.
- 22.7. No Director shall be entitled to appoint any person as an Alternate Director to himself.
- 22.8. In any election of Directors (other than appointment by the Board in terms of Article 22.4.2 above of a person in order to fill a casual vacancy which may arise from time to time or the election of such a person by the Members at a subsequent Members' Meeting, or the appointment of a Director by the Developer as contemplated in Article 22.3) the election is to be conducted at an Annual General Meeting, as follows
 - 22.8.1. The election is to be conducted in order to fill all vacancies on the Board at that time,
 - 22.8.2. The notice in respect of the Annual General Meeting shall set out which Directors are deemed to have, or will be deemed to have, retired and which of them have made themselves available for re-election.
 - 22.8.3. Any Member shall each be entitled to nominate 1 (one) person, including retired Directors who have made themselves available for re-election, to serve as a Director of the Association.
 - 22.8.4. All nominations are to reach the Association at least 30 (thirty) days before the date and time scheduled for the Annual General Meeting to begin. Each nominated person shall provide the Association with his consent in writing to serve as a Director, prior to the commencement of the Annual General Meeting.
 - 22.8.5. Every Member entitled to vote at the meeting may vote, for a number of candidates equal to the number of vacancies at that time, irrespective of the number of candidates appearing on the ballot paper.

- 22.8.6. Vacancies shall be filled by the candidates who have received the most votes. The aforementioned is illustrated as follows:
 - 22.8.6.1. In the event that there are 3 (three) vacancies, the three candidates who have received the top three amount of votes, shall be appointed.
- 22.9. No election of a Director shall take effect until he has delivered to the Association a written consent to serve and every candidate who failed to provide the Association with his written consent, shall be disqualified from the elections.
- 22.10. In the event of a vacancy arising on the Board, the continuing Directors (or sole continuing Director) may act notwithstanding any vacancy in their body, but, if and so long as their number is reduced below the minimum number fixed by or pursuant to this MOI, the continuing Directors or Director may act only for the purpose of summoning a Members' Meeting or procuring the election or appointment of interim Directors pursuant to Article 22.4.2.
- 22.11. If there is no Director able and willing to act, then any Member entitled to exercise voting rights in the election of a Director may convene a Members' Meeting for the purpose of appointing Directors.

23. CESSATION OF OFFICE AS DIRECTOR

- 23.1. The provisions of sections 69, 70 and 71 of the Act apply to the cessation of office as Director and, importantly, a Director shall cease to hold office as such
 - 23.1.1. immediately when he becomes Ineligible or Disqualified or the Board resolves to remove him on such basis, and in the latter case he has not within the permitted period filed an application for review or has filed such an application but the court has not yet confirmed the removal (during which period he shall be suspended);
 - 23.1.2. when he retires or is deemed to have retired;
 - 23.1.3. when he dies;
 - 23.1.4. when he resigns by written notice to the Association;
 - 23.1.5. if there are more than 3 (three) Directors in office and if the Board determines that he has become incapacitated to the extent that he is unable to perform the functions of a Director, and is unlikely to regain that capacity within a reasonable time, and the Director has not within the permitted period filed an application for

- review or has filed such an application but the court has not yet confirmed the removal (during which period he shall be suspended);
- 23.1.6. if he is declared delinquent by a court, or placed on probation under conditions that are inconsistent with continuing to be a Director of the Association;
- 23.1.7. if, without condonation of the Board, he is absent for 3 (three) consecutive meetings of the Board and the other Directors by majority decision declare that Director delinquent;
- 23.1.8. if he is removed by Special Resolution of the Members of the Association;
- 23.1.9. he files a petition for the surrender of his estate or an application for an administration order, or if he commits an act of insolvency as defined in the insolvency law for the time being in force, or if he makes any arrangement or composition with his creditors generally;
- 23.1.10. he is otherwise removed in accordance with any provisions of this MOI; or
- 23.1.11. if he was appointed as a Director by the Developer and the Developer, for whichever reason, ceases to be the registered owner of land in Monaghan Farm.

24. REMUNERATION OF DIRECTORS AND MEMBERS OF BOARD COMMITTEES

- 24.1. Directors and members of Board Committees, shall be entitled to be reimbursed all reasonable and *bona fide* expenses incurred by them during the performance of their duties as Directors and/or Chairman and/or Vice-Chairman and/or members of Board Committees, as the case may be.
- 24.2. Save as stated in Article 24, Directors and members of Board Committees shall not be entitled to any remuneration for the performance of their duties in terms hereof, unless the Members of the Association in an Annual General Meeting decide otherwise.
- 24.3. Notwithstanding any provision to the contrary in this MOI or elsewhere, the Directors and members of Board Committees shall not be entitled to pension benefits, gratuities or allowances.

25. FINANCIAL ASSISTANCE FOR DIRECTORS AND PRESCRIBED OFFICERS AND THEIR RELATED AND INTER-RELATED PARTIES

The Company shall not provide a loan to, secure a debt or obligation of, or otherwise provide direct or indirect financial assistance to, a director of the company or of a related or inter-related company, or to a person related to any such director.

26. GENERAL POWERS AND DUTIES OF DIRECTORS AND ELECTION OF CHAIRMAN AND VICE CHAIRMAN

26.1. The Directors may –

- 26.1.1. establish and maintain any non-contributory or contributory pension, superannuation, provident and benefit funds for the benefit of; and
- 26.1.2. give pensions, gratuities and allowances to and make payments for or towards the insurance of,

any persons who are employees or ex-employees (including in respect of indemnity insurance only, the Directors or ex-Directors) of the Association, or of any company which is or was a subsidiary of the Association or is or was in any way allied to or associated with it or any such subsidiary, and the wives, widows, families and dependants of such persons.

- 26.2. The Board shall be entitled to annually elect from its members a Chairman and a Vice-Chairman and determine the period for which they are to hold office, provided that the offices of the Chairman and Vice-Chairman shall *ipso facto* be vacated by the Director holding such office upon his ceasing to be a Director for any reason.
- 26.3. The Chairman shall preside at all meetings of the Board and all Members' Meetings and shall perform all duties incidental to the office of the Chairman and such other duties as may be prescribed by the Board from time to time. In the event of an equality of votes, the Chairman shall not have a second and/or casting vote.
- 26.4. The Vice-Chairman shall assume the powers and duties of the Chairman in the absence of the Chairman or his inability or refusal to act as Chairman, and failing the appointment of a Vice Chairman, a Director nominated by the majority of the remaining Directors of the Board shall perform such duties as may from time to time be assigned to him by the Chairman of the Board.
- 26.5. The Board may in its discretion from time to time for the purposes of the Association borrow or raise such sums of money from Members of the Association or from such other persons as the Board may determine. The Board may (but shall not be obliged to) further secure the repayment of any sums so raised in such a manner and upon such terms as it may deem fit, whether by the mortgaging of fixed property belonging to or in respect of which the Association has a mortgageable interest or otherwise.

- 26.6. The Board shall have the right to vary, cancel or modify any of its decisions and resolutions from time to time.
- 26.7. The Board (and each individual Member of the Board) shall be entitled, but not obliged, -
 - 26.7.1. to engage, on behalf of the Association, the services of accountants, auditors, attorneys, advocates, architects, engineers and/or any other profession, and
 - 26.7.2. to obtain opinions and/or other professional services from such persons relating to the business of the Association only;

as and when the Board (or the individual Director) may deem necessary and/or required, and on such terms as the Board (or the individual Director) may decide. In the event that the Board (or an individual Director), engages such professional persons as aforesaid, the Association shall be liable for payment of such professional persons' accounts.

- 26.8. The Board shall further have the power
 - 26.8.1. To require that any works being constructed within Monaghan Farm be supervised to ensure that the provisions of this MOI, the rules and/or regulations of the Association, are complied with and that all work is performed in a proper and workmanlike manner.
 - 26.8.2. To issue architectural design manuals in respect of Monaghan Farm.
- 26.9. The Board shall be entitled, whenever it considers that the appearance of any land or building within Monaghan Farm and owned by a Member, is unsightly or injurious to the amenities of the surrounding area of Monaghan Farm generally, to serve notice on such Member requiring him to take steps to rectify such unsightly or injurious condition within a stated period.
- 26.10. Should any Member, on whom notice as contemplated in Article 26.9 has been served, fail to take such steps as may be specified in the notice within the stated period, the Board shall be entitled, but not obliged, on behalf of the Association, to take such steps as may be reasonably required and/or necessary, including legal steps, in order to rectify such unsightly or injurious condition and to recover the costs of doing so from the Member concerned, which costs shall be deemed to be a debt owing by such Member to the Association.
- 26.11. The Board may from time to time determine the routine maintenance requirements of the common property and instruct a person to attend to such maintenance requirements on behalf of the Association and at its cost.

26.12. The Board shall be obliged to maintain such measures as it deems necessary, in its sole discretion, to ensure the security access to and safety of Monaghan Farm and all persons lawfully therein.

27. BOARD COMMITTEES

- 27.1. The Directors may appoint any number of Board committees for any period, or once off for a specific purpose, and delegate to such committees any authority of the Board. The members of such committees may include persons who are not Directors, as long as they are not Ineligible or Disqualified to be Directors, who shall not be able to vote at Board meetings.
- 27.2. No person shall be appointed as a member of a Board committee, if he is Ineligible or Disqualified and any such appointment shall be a nullity. A person who is Ineligible or Disqualified must not consent to be appointed as a member of a Board committee nor act as such a member. A person placed under probation by a court must not serve as a member of a Board committee unless the order of court so permits.
- 27.3. There are no general qualifications prescribed by the Association for a person to serve as a member of a Board committee in addition to the requirements of the Act.
- 27.4. A Member of a Board committee shall cease to hold office as such immediately he becomes Ineligible or Disqualified in terms of the Companies Act.
- 27.5. Committees of the Board may consult with or receive advice from any person, including professional persons. Insofar as may be permissible by the Act, the members of Board Committees shall be entitled to vote at such committee meetings on matters to be decided by such Committee.
- 27.6. Meetings and other proceedings of a committee of the Board consisting of more than 1 (one) Member shall be governed by the provisions of this MOI regulating the meetings and proceedings of Directors.

28. PERSONAL FINANCIAL INTERESTS OF DIRECTORS

- 28.1. For the purposes of this Article 28 (*Personal Financial Interests of Directors*), "Director" includes a Prescribed Officer and a person who is a Member of a committee of the Board, irrespective of whether or not the person is also a Member of the Board.
- 28.2. At any time, a Director may disclose any personal financial interest in advance, by delivering to the Board a notice in Writing setting out the nature and extent of that personal financial

- interest, to be used generally by the Association until changed or withdrawn by further Written notice from that Director.
- 28.3. If a Director has a personal financial interest in respect of a matter to be considered at a meeting of the Board, or knows that a related person has a personal financial interest in the matter, the Director
 - 28.3.1. must disclose the personal financial interest and its general nature before the matter is considered at the meeting;
 - 28.3.2. must disclose to the meeting any material information relating to the matter, and known to the Director;
 - 28.3.3. may disclose any observations or pertinent insights relating to the matter if requested to do so by the other Directors;
 - 28.3.4. if present at the meeting, must leave the meeting immediately after making any disclosure contemplated in Articles 28.3.2 or 28.3.3;
 - 28.3.5. must not take part in the consideration of the matter, except to the extent contemplated in Articles 28.3.2 or 28.3.3;
 - 28.3.6. while absent from the meeting in terms of this Article 28.3:
 - 28.3.6.1. is to be regarded as being present at the meeting for the purpose of determining whether sufficient Directors are present to constitute a quorum; and
 - 28.3.6.2. is not to be regarded as being present at the meeting for the purpose of determining whether a resolution has sufficient support to be adopted; and
 - 28.3.7. must not execute any document on behalf of the Association in relation to the matter unless specifically requested or directed to do so by the Board.
- 28.4. If a Director acquires a personal financial interest in an agreement or other matter in which the Association has a material interest, or knows that a related person has acquired a personal financial interest in the matter, after the agreement or other matter has been approved by the Association, the Director must promptly disclose to the Board the nature and extent of that personal financial interest, and the material circumstances relating to the Director or related person's acquisition of that personal financial interest.

- 28.5. A decision by the Board, or a transaction or agreement approved by the Board is valid despite any personal financial interest of a Director or person related to the Director, only if
 - 28.5.1. it was approved following the disclosure of the personal financial interest in the manner contemplated in this Article 27; or
 - 28.5.2. despite having been approved without disclosure of that personal financial interest, it has been ratified by an Ordinary Resolution by the Members following disclosure of that personal financial interest or so declared by a court.
- 28.6. Disclosure of personal financial interests in terms of this Article 28.3 above, does not apply to a Director
 - 28.6.1. in respect of a decision that may generally affect -
 - 28.6.1.1. all of the Directors of the Association in their capacity as directors;
 - 28.6.1.2. a class of persons, despite the fact that the Director is one member of that classs of persons, unless the only members of that class are the director or persons related or inter-related to the director;
 - 28.6.2. In respect of a proposal to remove that Director from office as contemplated in section 71 of the Act.

29. PROCEEDINGS OF DIRECTORS

- 29.1. A Director authorised by the Board
 - 29.1.1. may, at any time, summon a meeting of the Directors; and
 - 29.1.2. must call a meeting of the Directors if required to do so by at least 2 (two) Directors.
- 29.2. The Directors may determine what period of notice, being not less than 48 (forty eight) hours, shall be given of meetings of Directors and may determine the means of giving such notice, in writing, which may include telephone, fax or Electronic Communication. It shall be necessary to give notice of a meeting of Directors to all Directors even those for the time being absent from South Africa.
- 29.3. If all of the Directors
 - 29.3.1. acknowledge actual receipt of the notice; or
 - 29.3.2. are present at a meeting of the Directors; or

29.3.3. waive notice of the meeting,

the meeting may proceed even if the Association failed to give the required notice of that meeting, or there was a defect in the giving of the notice.

- 29.4. The Directors may meet together for the despatch of business, adjourn and otherwise regulate their meetings as they think fit provided that they shall meet, as a minimum, at least once every quarter. Unless otherwise resolved by the Board, all Board meetings shall take place at Monaghan Farm.
- 29.5. A meeting of Directors may be conducted by Electronic Communication and/or one or more Directors may participate in a meeting of Directors by Electronic Communication so long as the Electronic Communication facility employed ordinarily enables all persons participating in that meeting to communicate concurrently with each other without an intermediary, and to participate effectively in the meeting.
- 29.6. The quorum for a Directors' meeting is established when a majority of the serving Directors are personally present: Provided that, should a Director attend a Board meeting through electronic communication, that Director shall be deemed to have been present at the meeting. In addition, a Director who has authorised another Director, in writing, to attend a meeting and vote thereat on his or behalf, shall be deemed to be present himself.
- 29.7. The Chairman shall preside as such at all meetings of the Board; Provided that should, at any meeting of the Board, the Chairman not be present within 15 (fifteen) minutes after the time appointed for the holding thereof, then the Vice-Chairman shall act as Chairman at such meeting, provided further that should the Vice-Chairman also not be present within 15 (fifteen) minutes of the time appointed for the holding of such a meeting, those Directors present shall appoint a Chairman from the meeting, who shall thereupon exercise all the powers and duties of the Chairman in relation to that meeting.
- 29.8. If within 30 (thirty) minutes from the time appointed for the holding of a Board meeting a quorum is not present, the meeting shall stand adjourned to the same day in the next week at the same place and time, or at such other place as the Chairman shall appoint, and if at such subsequent meeting the subject to the adjournment, a quorum is not present within half an hour from the time appointed for holding the meeting, the Directors present shall be a quorum.
- 29.9. Each Director has 1 (one) vote on a matter before the Board and a simple majority of the votes cast on a resolution is sufficient to approve that resolution.

- 29.10. In the event of an equality of votes for and against any resolution, the Chairman will not have a second or casting vote.
- 29.11. The Association must keep minutes of the meetings of the Board, and any of its committees, and include in the minutes
 - 29.11.1. any declaration personal financial interest;
 - 29.11.2. every resolution adopted;.
- 29.12. Resolutions of the Board are made by simple majority and --
 - 29.12.1. must be dated and sequentially numbered; and
 - 29.12.2. are effective as of the date of the resolution, unless the resolution states otherwise, subject to Article 28.13 below.
- 29.13. Any minutes of a meeting, or a resolution, signed by the Chairman of the meeting, or by the Chairman of the next meeting of the Board, are/is evidence of the proceedings of that meeting, or adoption of that resolution, as the case may be.
- 29.14. A resolution placed before, or submitted to, the Directors outside of a formal Board meeting and signed by all Directors shall be referred to as a Round Robin Resolution of the Directors and be valid in all respects as if it had been duly passed at a meeting of the Board and shall form part of the records of the next Board meeting.

30. COMMON PROPERTY

- 30.1. The common property in the case of a sectional title scheme within Monaghan Farm, vests in the body corporate of that scheme subject to the powers, obligations and Rules of the Association.
- 30.2. All common property which vests in the Association is subject to the provisions of this MOI and the Rules.
- 30.3. Members shall be entitled access to all common property in respect of which the Association holds title or tenancy subject to compliance by Members with the Rules pertaining to such common property.
- 30.4. The Board may from time to time limit, restrict or suspend access in relation to any such part of the common property, as may, in their sole discretion, be necessary and/or required.
- 30.5. The Board may make rules relating to access by the general public to certain common property in order to confine the public access to such areas.
- 30.6. The Board may require any Member to maintain the sidewalk adjacent to his property and in the event of such Member failing to maintain such sidewalk to the satisfaction of the Board, the Board shall be entitled to take such action as may be necessary for the maintenance of such sidewalk and to charge the Member concerned.
- 30.7. Where the boundary of any one Member's property also constitutes the boundary of Monaghan Farm, such Member shall be obliged to permit the Association to erect upon such Member's property immediately adjacent to such boundary, such walling or other fencing as the Board may determine. Such Member shall not be entitled to interfere in any manner whatsoever with any such walling or fencing, and shall permit the Association from time-to-time access to such Member's property in order to inspect such walling or fencing and to effect such repairs as may be necessary from time to time. In the event of such Member wishing to erect his own walling or fencing, he shall be obliged to erect same within the walling or fencing referred to above and subject to such conditions as the Board may lay down to permit the Association access to the boundary walling or fencing concerned.

31. ALIENATION

31.1. No Member shall alienate land or rights thereto until a Director, on the Board's behalf, has certified that the Member has at date of transfer fulfilled all his financial obligations to the Association.

- 31.2. No land or any interest therein shall further be alienated without the consent of the Association, which consent shall not be unreasonably withheld
 - 31.2.1. unless such Member is indebted to the Association in any way in respect of levy, special and/or additional contributions, fines or other penalties, interest or any other debt due which the Association may in terms of this MOI be entitled to claim from him; including the payment of the re-sale fee, unless the payment of the resale fee is excempted or the re-sale fee has been secured to the satisfaction of the Board of the Association
 - 31.2.2. the beneficiary of the proposed alienation has agreed to bind himself, to the satisfaction of the Board as a contract to the benefit of the Association, to become a Member of the Association and the commitment thereto is registered on the title deeds of the land.
- 31.3. Before land is transferred, the Member who has alienated such land shall pay to the Association an administration fee for issuing a clearance certificate in respect of the proposed transfer as may be determined by the Board from time to time and any alienation shall take place prior to or on the date specified in the clearance certificate.
- 31.4. Any registration of transfer of land into the name of the transferee shall *ipso facto* constitute the transfer of Membership to the transferee.

32. LOSS OF DOCUMENTS

The Association shall not be responsible for the loss in transmission of any cheque, warrant, certificate or (without any limitation *eiusdem generis*) other document sent through the post either to the registered address of any Member or to any other address requested by the Member.

33. NOTICES

- 33.1. The Association may give notices, documents, records or statements or notices of availability of the aforegoing by personal delivery to the Member or by sending them prepaid through the post or by transmitting them via Electronic Communication.
- 33.2. Any Member who/which has furnished an electronic address to the Association, by doing so
 - 33.2.1. authorises the Association to use Electronic Communication to give notices, documents, records or statements or notices of availability of the aforegoing to him; and

- 33.2.2. confirms that same can conveniently be printed by the Member within a reasonable time and at a reasonable cost.
- 33.3. Any notice required to be given by the Association to the Members, and not expressly prohibiting the provisions of this Article from applying, shall be sufficiently given (subject to giving a notice of availability in accordance with Articles 33.1 or 33.2), if given by posting it on the Association's web site until at least the date when the event to which the notice refers occurs.
- 33.4. Any notice, document, record or statement or notice of availability of the aforegoing sent by the Association shall be deemed to have been delivered on the date and time determined in accordance with Schedule 3 hereto.
- 33.5. A Member shall be bound by every notice Delivered to the person who was, at the date on which that notice was Delivered or established to the satisfaction of the Directors (as the case may be) as the Member, notwithstanding that the Member may then have been dead or may subsequently have died or have been or become otherwise incapable of acting as a Member.
- 33.6. The Association shall not be bound to use any method of giving notice, documents, records or statements or notices of availability of the aforegoing, contemplated in the Regulations in respect of which provision is made for deemed delivery, but if the Association does use such a method, the notice, document, record or statement or notice of availability of the aforegoing shall be deemed to be delivered on the day determined in accordance with the Regulations.
- 33.7. As regards the signature of an Electronic Communication by a Member, it shall be in such form as the Directors may specify to demonstrate that the Electronic Communication is genuine, or failing any such specification by the Directors, it shall be constituted by the Member indicating in the Electronic Communication that it is the Member's intention to use the Electronic Communication as the medium to indicate the Member's approval of the information in, or the Member's signature of the document in or attached to, the Electronic Communication which contains the name of the Member sending it in the body of the Electronic Communication.

34. INDEMNITY

34.1. For the purposes of this Article 34, "Director" includes a former Director, a Prescribed Officer, a person who is a member of a committee of the Board, irrespective of whether or not the person is also a member of the Board.

34.2. The Association may –

- 34.2.1. not directly or indirectly pay any fine that may be imposed on a Director, or on a Director of a related company, as a consequence of that Director having been convicted of an offence in terms of any national legislation, unless the conviction was based on strict liability;
- 34.2.2. advance expenses to a Director to defend litigation in any proceedings arising out of the Director's service to the Association; and
- 34.2.3. directly or indirectly indemnify a Director for -
 - 34.2.3.1. any liability, other than in respect of –
 - 34.2.3.2. any liability arising in terms of Section 77(3)(a), (b) or (c) of the Companies Act or from wilful misconduct or wilful breach of trust on the part of the Director; or
 - 34.2.3.3. any fine contemplated in Article 33.2.1;
 - 34.2.3.4. any expenses contemplated in Article 33.2.2 irrespective of whether it has advanced those expenses, if the proceedings
 - 34.2.3.4.1. are abandoned or exculpate the Director; or
 - 34.2.3.4.2. arise in respect of any other liability for which the Association may indemnify the Director in terms of Article 28.2.3.1.
- 34.3. The Association may purchase insurance to protect
 - 34.3.1. a Director against any liability or expenses contemplated in Article 28.2.2 or 28.2.3; or
 - 34.3.2. the Association against any contingency including but not limited to -
 - 34.3.2.1. any expenses -
 - 34.3.2.2. that the Association is permitted to advance in accordance with Article 28.2.2; or
 - 34.3.2.3. for which the Association is permitted to indemnify a Director in accordance with Article 33.2.3.2; or

- 34.3.2.4. any liability for which the Association is permitted to indemnify a Director in accordance with Article 33.2.3.1.
- 34.4. The Association is entitled to claim restitution from a Director or of a related company for any money paid directly or indirectly by the Association to or on behalf of that Director in any manner inconsistent with Section 75 of the Act.
- 34.5. Each and every Member individually shall indemnify and hold harmless the Association against any action by the Member, members of his family or any person within Monaghan Farm at the invitation of or under the control of the Member concerned, whatsoever the nature of such claims and howsoever arising.
- 34.6. Any person using any of the services, common property and/or facilities of and in Monaghan Farm does so entirely at his own risk and no person shall have any claim against the Association of whatsoever nature arising from such use, nor for anything which may befall any person during the course of such use, whether caused by human or animal agency, natural phenomena or otherwise.
- 34.7. No person or Member shall have any claim against the Association as a result of fire or any substance, liquid or gas, escaping from the property of the Association and causing damage to any person or property of any Member.

35. ACCEPTANCE OF CESSIONS OF RIGHTS

- 35.1. The Association shall accept and assume in every instance all rights ceded to it in writing arising from the Sales Agreements entered into between a Member and the Developer and enduring beyond the completion of the first transfer of the erf or unit by the Developer in respect of:
 - 35.1.1. the rights of the Developer to grant or withhold approvals relating to the subsequent use of that erf or unit; and
 - 35.1.2. the rights of the Purchaser relating to the provision by the Developer of common facilities intended for the benefit and enjoyment of all members;

and the Association shall act upon those rights in every instance in a manner consistent with the objectives of the Association and to the collective benefit of its members.

36. ACCEPTANCE OF COMMON FACILITIES

36.1. The Developer undertakes to cause to be delivered to the Association the common facilities further to those transferred and/or completed as at the date of adoption of this MOI being

- 36.1.1. two tennis courts
- 36.1.2. one squash court
- 36.1.3. one gymnasium
- 36.1.4. 30 stables and equestrian facilities
- 36.2. The Board may at any time require of the Developer that construction of any of the abovementioned common facilities be initiated whereupon the Developer shall, within six months, submit all necessary applications for approvals to the relevant authorities and upon receipt of which construction shall commence without unreasonable delay.
- 36.3. Upon completion of each of the common facilities, the Board shall have the right to require that the common facility be in compliance with the applicable municipal standard, if any, and certified as fit-for-purpose by the relevant architect, engineers(or contractors) whereupon the Association shall, upon request of the Developer, be obliged to take transfer of that common facility in title and/or responsibility for its maintenance and use.
- 36.4. Following recommendation by the Board, adoption by Special Resolution of the Members, and agreement with the Developer, the nature, number or extent of any of the common facilities may without limitation be amended, substituted, or exchanged save that the consequence of such amendment, substitution, or exchange shall not be inconsistent with the objectives of the Association and shall not result in any substantive diminution of overall benefit to its members.
- 36.5. The Association may, in the alternative and prior to the commencement of construction of any common facility elect to accept payment from the Developer of an agreed amount in respect and in lieu of that common facility, provided that
 - 36.5.1. within twelve months, those funds are applied to the provision of similar or other recreational facilities of comparable longevity, and
 - 36.5.2. following receipt in full of that agreed amount, the Developer shall be released from its obligation to cause delivery of that common facility to the Association.
- 36.6. To the extent, if any, that the Developer intends to enter into a tenancy agreement respect of any common facility prior to its delivery to the Association, the Developer shall:
 - 36.6.1. procure the agreement of the Board as to the terms and conditions of that tenancy agreement and, if applicable, the agreement of the Board to assume the rights and

- responsibilities of the Developer under the tenancy agreement upon a transfer of title or responsibility for that common facility to the Association; or
- 36.6.2. provide for the termination of that tenancy agreement upon transfer of title or responsibility for that common facility to the Association.

ANNEXURE 1

Definitions in the Companies Act

"accounting records" means information in written or electronic form concerning the financial affairs of a company as required in terms of this Act including, but not limited to, purchase and sales records, general and subsidiary ledgers and other documents and books used in the preparation of financial statements; ¹

"alternate director" means a person elected or appointed to serve, as the occasion requires, as a member of the board of a company in substitution for a particular elected or appointed director of that company;

"annual general meeting" means the meeting of a public company required by section 61(7);

"audit" has the meaning set out in the Auditing Profession Act, but does not include an "independent review" of annual financial statements, as contemplated in section 30(2)(b)(ii)(bb);

"auditor" has the meaning set out in the Auditing Profession Act;

"Banks Act" means the Banks Act, 1990 (Act No. 1194 of 1990);

"board" means the board of directors of a company;

"business days" has the meaning determined in accordance with section 5(3);

"Commission" means the Companies and Intellectual Property Commission established by section 185;

"Commissioner" means the person appointed to or acting in the office of that name, as contemplated in Section 189;

"company" means a juristic person incorporated in terms of this Act, a domesticated company, or a juristic person that, immediately before the effective date—

- (a) was registered in terms of the—
 - (i) Companies Act, 1973 (Act No. 61 of 1973), other than as an external company as defined in that Act; or
 - (ii) Close Corporations Act, 1984 (Act No. 69 of 1984), if it has subsequently been converted in terms of 0;

¹ Regulation 25(3) contains requirements as to what the accounting records must include.

- (b) was in existence and recognised as an 'existing company' in terms of the Companies Act, 1973 (Act No. 61 of 1973); or
- (c) was deregistered in terms of the Companies Act, 1973 (Act No. 61 of 1973), and has subsequently been re-registered in terms of this Act;

"Competition Act", means the Competition Act, 1998 (Act No. 89 of 1998);

"convertible" when used in relation to any securities of a company, means securities that may, by their terms, be converted into other securities of the company, including—

- (a) any non-voting securities issued by the company and which will become voting securities—
 - (i) on the happening of a designated event; or
 - (ii) if the holder of those securities so elects at some time after acquiring them; and
- (b) Options to acquire securities to be issued by the company, irrespective of whether those securities may be voting securities, or non-voting securities contemplated in paragraph (a);

"creditor" means a person to whom a company is or may become obligated in terms of any liability or other obligation that would be required to be considered by the company if it were applying the solvency and liquidity test set out in section 4;

"director" means a member of the board of a company, as contemplated in section 66, or an alternate director of a company and includes any person occupying the position of a director or alternative director, by whatever name designated;

"effective date", with reference to any particular provision of this Act, means the date on which that provision came into operation in terms of section 225;

"electronic communication" has the meaning set out in section 1 of the Electronic Communications and Transactions Act;

"Electronic Communications and Transactions Act" means the Electronic Communications and Transactions Act, 2002 (Act No. 25 of 2002);

"employee share scheme" has the meaning set out in section 95(1)(c);

"exchange" when used as a noun, has the meaning set out in section 1 of the Securities Services Act, 2004 (Act No. 36 of 2004);

"exercise", when used in relation to voting rights, includes voting by proxy, nominee, trustee or other person in a similar capacity;

"ex officio director" means a person who holds office as a director of a particular company solely as a consequence of that person holding some other office, title, designation or similar status specified in the company's Memorandum of Incorporation;

"external company" means a foreign company that is carrying on business, or non-profit activities, as the case may be, within the Republic, subject to section 23(2);

"financial statement" includes—

- (a) annual financial statements and provisional annual financial statements;
- (b) interim or preliminary reports;
- (c) group and consolidated financial statements in the case of a group of companies; and
- (d) financial information in a circular, prospectus or provisional announcement of results, that an actual or prospective creditor or holder of the company's securities, or the Commission, Panel or other regulatory authority, may reasonably be expected to rely on;

"group of companies" means a holding company and all of its subsidiaries;

"holding company", in relation to a subsidiary, means a juristic person that controls that subsidiary as a result of any circumstances contemplated in Section 2(2)(a) or 3(1)(a);

"incorporator", when used—

- (a) with respect to a company incorporated in terms of this Act, means a person who incorporated that company, as contemplated in Section 13; or
- (b) with respect to a pre-existing company, means a person who took the relevant actions comparable to those contemplated in Section 13 to bring about the incorporation of that company;

"individual" means a natural person;

"inter-related", when used in respect of three or more persons, means persons who are related to one another in a linked series of relationships, such that two of the persons are related in a manner contemplated in Section 2(1) and one of them is related to the third in any such manner, and so forth in an unbroken series:

"juristic person" includes—

- (a) a foreign company; and
- (b) a trust, irrespective of whether or not it was established within or outside the Republic;

"knowing", "knowingly" or "knows", when used with respect to a person, and in relation to a particular matter, means that the person either —

- (a) had actual knowledge of the matter; or
- (b) was in a position in which the person reasonably ought to have
 - (i) had actual knowledge;
 - (ii) investigated the matter to an extent that would have provided the person with actual knowledge; or
 - (iii) taken other measures which, if taken, could reasonably be expected to have provided the person with actual knowledge of the matter;

"nominee" has the meaning set out in Section 1 of the Securities Services Act, 2004 (Act No. 36 of 2004);

"ordinary resolution" means a resolution adopted with the support of more than 50% of the voting rights exercised on the resolution, or a higher percentage as contemplated in Section 65(8) at a members' meeting.

"person" includes a juristic person;

"personal financial interest", when used with respect to any person—

- (a) means a direct material interest of that person, of a financial, monetary or economic nature, or to which a monetary value may be attributed; but
- (b) does not include any interest held by a person in a unit trust or collective investment scheme in terms of the Collective Investment Schemes Act, 2002 (Act No. 45 of 2002), unless that person has direct control over the investment decisions of that fund or investment;

"prescribed officer" means a person who, within a company, performs any function that has been designated by the Minister in terms of Section 66(10);

"present at a meeting" means to be present in person, or able to participate in the meeting by electronic communication, or to be represented by a proxy who is present in person or able to participate in the meeting by electronic communication;

"public company" means a profit company that is not a state-owned company, a private company or a personal liability company;

"record date" means the date established under Section 59 on which a company determines the identity of its shareholders and their shareholdings for the purposes of this Act;

"registered office" means the office of a company, or of an external company, that is registered as required by Section 23;

"related", when used in respect of two persons, means persons who are connected to one another in any manner contemplated in Section 2(1)(a) to Section (c);

"rules" and "rules of a company" means any rules made by a company as contemplated in Section 15(3) to (5);

"special resolution" means —

- (a) in the case of a company, a resolution adopted with the support of at least 75% of the voting rights exercised on the resolution, or a different percentage as contemplated in section 65(10) at a members' meeting; or
- (b) in the case of any other juristic person, a decision by the owner or owners of that person, or by another authorized person, that requires the highest level of support in order to be adopted, in terms of the relevant law under which that juristic person was incorporated;

"subsidiary" has the meaning determined in accordance with Section 3;

"wholly-owned subsidiary" has the meaning determined in accordance with Section 3(1)(b).

ANNEXURE 2

Directors - Ineligible / disqualified in terms of Section 69(7) and (8) of the Companies Act read with Regulation 39(3)

A person is ineligible to be a Director if the Person —

is a juristic person;

is an unemancipated minor, or is under a similar legal disability; or

does not satisfy any qualification set out in the MOI.

A person is disqualified to be a Director if --

a court has prohibited that Person to be a Director, or declared the Person to be delinquent in terms of Section 162, or in terms of Section 47 of the Close Corporations Act, 1984 (Act No. 69 of 1984); or

the Person --

is an unrehabilitated insolvent;

is prohibited in terms of any public regulation to be a Director;

has been removed from an office of trust, on the grounds of misconduct involving dishonesty; or

has been convicted, in the Republic or elsewhere, and imprisoned without the option of a fine, or fined more than R1 000,00 (one thousand rand) amount, for theft, fraud, forgery, perjury or an offence —

involving fraud, misrepresentation or dishonesty;

in connection with the promotion, formation or management of a company, or in connection with any act contemplated in subsection (2) or (5); or

under the Companies Act, the Insolvency Act, 1936 (Act No. 24 of 1936), the Close Corporations Act, 1984, the Competition Act, the Financial Intelligence Centre Act, 2001 (Act No. 38 of 2001), the Securities Services Act, 2004 (Act No. 36 of 2004), or Chapter 2 of the Prevention and Combating of Corruption Activities Act, 2004 (Act No. 12 of 2004).

ANNEXURE 3

Prescribed Methods of Delivery in the Regulations

Person to	who	om	the
document	is	to	be
delivered			

Method of delivery

Date and Time of Deemed delivery

Any Person

By faxing the notice or a certified copy of the document to the Person, if the Person has a fax number:

On the date and at the time recorded by the fax receiver, unless there is conclusive evidence that it was delivered on a different date or at a different time

By sending the notice or a copy of the document by electronic mail, if the Person has an Electronic Address:

On the date and at the time recorded by the computer used by the Company, unless there is conclusive evidence that it was delivered on a different date or at a different time.

By sending the notice or a certified copy of the document by registered post to the Person's last known address;

On the 7th (seventh) day following the day on which the notice or document was posted as recorded by a post office, unless there is conclusive evidence that it was delivered on a different day.

By any other means authorised by the High Court; or

In accordance with the order of the High Court.

By any other method allowed for that Person in terms of the following rows of this Table.

As provided for that method of delivery.

Any natural Person

certified copy of the document to Person. the or to any representative authorised writing to accept service on behalf of the Person:

By handing the notice or a On the date and at the time recorded on a receipt for the delivery.

certified copy of the document at on a receipt for the delivery. the Person's place of residence or business with any other Person who is apparently at least 16 (sixteen) years old and in charge of the premises at the time;

By leaving the notice or a On the date and at the time recorded

certified copy of the document at Person's place employment with any Person who is apparently at least 16 (sixteen) years old and apparently in authority.

By leaving the notice or a On the date and at the time recorded on a receipt for the delivery.

A company or similar body corporate

By handing the notice or a certified copy of the document to a responsible employee of the company or body corporate at its registered office or its principal place of business within South Africa:

On the date and at the time recorded on a receipt for the delivery.

If there is no employee willing to accept service, by affixing the notice or a certified copy of the document to the main door of the office or place of business.

On the date and at the time sworn to by affidavit of the Person who affixed the document, unless there is conclusive evidence that the document was affixed on a different date or at a different time.

The state or province

a By handing the notice or a certified copy of the document to a responsible employee in any office of the State Attorney.

On the date and at the time recorded on a receipt for the delivery.

A municipality

By handing the notice or a certified copy of the document to the town clerk, assistant town clerk or any Person acting on behalf of that Person.

On the date and at the time recorded on a receipt for the delivery.

A trade union

By handing the notice or a certified copy of the document to a responsible employee who is apparently in charge of the main office of the union.

On the date and at the time recorded on a receipt for the delivery.

If there is no person willing to accept service, by affixing a certified copy of the notice or document to the main door of that office.

On the date and at the time sworn to by affidavit of the Person who affixed the document, unless there is conclusive evidence that the document was affixed on a different date or at a different time.

Employees of the Company

By fixing the notice or certified copy of the document, in a prominent place in the workplace where it can be easily read by employees. On the date and at the time sworn to by affidavit of the Person who affixed the document, unless there is conclusive evidence that the document was affixed on a different date or at a different time.

A partnership, firm or association

By handing the notice or a certified copy of the document to a Person who is apparently in charge of the premises and apparently at least 16 (sixteen) years of age, at the place of business of the partnership, firm or association;

On the date and at the time recorded on a receipt for the delivery.

association has no place of on a receipt for the delivery. business, by handing the notice or a certified copy of the document to a partner, the owner of the firm, or the chairman or secretary of the managing or other controlling body of the association, as the case may be.

If the partnership, firm or On the date and at the time recorded

ANNEXURE 4

SCHEDULE 1 TO THE COMPANIES ACT

PROVISIONS CONCERNING NON-PROFIT COMPANIES

- 1 Objects and policies
 - (1) The Memorandum of Incorporation of a non-profit company must-
 - (a) set out at least one object of the company, and each such object must be either-
 - (i) a public benefit object; or
 - (ii) an object relating to one or more cultural or social activities, or communal or group interests; and
 - (b) be consistent with the principles set out in sub-items (2) to (9).
 - (2) A non-profit company-
 - (a) must apply all of its assets and income, however derived, to advance its stated objects, as set out in its Memorandum of Incorporation; and
 - (b) subject to paragraph (a), may-
 - (i) acquire and hold securities issued by a profit company; or
 - (ii) directly or indirectly, alone or with any other person, carry on any business, trade or undertaking consistent with or ancillary to its stated objects.
 - (3) A non-profit company must not, directly or indirectly, pay any portion of its income or transfer any of its assets, regardless how the income or asset was derived, to any person who is or was an incorporator of the company, or who is a member or director, or person appointing a director, of the company, except-
 - (a) as reasonable-
 - (i) remuneration for goods delivered or services rendered to, or at the direction of, the company; or

- (ii) payment of, or reimbursement for, expenses incurred to advance a stated object of the company;
- (b) as a payment of an amount due and payable by the company in terms of a bona fide agreement between the company and that person or another;
- (c) as a payment in respect of any rights of that person, to the extent that such rights are administered by the company in order to advance a stated object of the company; or
- (d) in respect of any legal obligation binding on the company.
- (4) Despite any provision in any law or agreement to the contrary, upon the winding-up or dissolution of a non-profit company-
 - (a) no past or present member or director of that company, or person appointing a director of that company, is entitled to any part of the net value of the company after its obligations and liabilities have been satisfied; and
 - (b) the entire net value of the company must be distributed to one or more non-profit companies, registered external non-profit companies carrying on activities within the Republic, voluntary associations or non-profit trusts-
 - (i) having objects similar to its main object; and
 - (ii) as determined-
 - (aa) in terms of the company's Memorandum of Incorporation;
 - (bb) by its members, if any, or its directors, at or immediately before the time of its dissolution; or
 - (cc) by the court, if the Memorandum of Incorporation, or the members or directors fail to make such a determination.
- (5) The Commission may apply to the court, on behalf of a non-profit company, for a determination contemplated in sub-item (4) (b) (ii) (cc) if the non-profit company has-
 - (a) no remaining members or directors; and
 - (b) failed to-

- (i) make a determination contemplated in sub-item (4) (b) (ii) (bb); or
- (ii) apply to the court for such a determination.
- (6) Incorporation as a non-profit company in terms of this Act, or registration as an external non-profit company in terms of this Act, and compliance by either with the provisions of this Act does not necessarily qualify that non-profit company, or external non-profit company, for any particular status, category, classification or treatment in terms of the Income Tax Act, 1962 (Act 58 of 1962), or any other legislation, except to the extent that any such legislation provides otherwise.
- (7) Each voting member of a non-profit company has at least one vote.
- (8) The vote of each member of a non-profit company is of equal value to the vote of each other voting member on any matter to be determined by vote of the members, except to the extent that the company's Memorandum of Incorporation provides otherwise.
- (9) If a non-profit company has members, the requirement in section 24 (4) to maintain a securities register must be read as requiring the company to maintain a membership register.

2 Fundamental transactions

- (1) A non-profit company may not-
 - (a) amalgamate or merge with, or convert to, a profit company; or
 - (b) dispose of any part of its assets, undertaking or business to a profit company, other than for fair value, except to the extent that such a disposition of an asset occurs in the ordinary course of the activities of the non-profit company.
- (2) If a non-profit company has voting members, any proposal to-
 - (a) dispose of all or the greater part of its assets or undertaking; or
- (b) amalgamate or merge with another non-profit company, must be submitted to the voting members for approval, in a manner comparable to that required of profit companies in accordance with sections 112 and 113, respectively.
- (3) Sections 115 and 116, read with the changes required by the context, apply with respect to the approval of a proposal contemplated in sub-item (2).

3 Incorporators of non-profit company

The incorporators of a non-profit company are its-

- (a) first directors; and
- (b) its first members, if its Memorandum of Incorporation provides for it to have members.

4 Members

- (1) A non-profit company is not required to have members, but its Memorandum of Incorporation may provide for it to do so.
- (2) If the Memorandum of Incorporation of a non-profit company provides for the company to have members, it-
 - (a) must not restrict or regulate, or provide for any restriction or regulation of, that membership in any manner that amounts to unfair discrimination in terms of section 9 of the Constitution:
 - (b) must not presume the membership of any person, regard a person to be a member, or provide for the automatic or ex officio membership of any person, on any basis other than life-time membership awarded to a person-
 - (i) for service to the company or to the public benefit objects set out in the company's Memorandum of Incorporation; and
 - (ii) with that person's consent;
 - (c) may allow for membership to be held by juristic persons, including profit companies;
 - (d) may provide for no more than two classes of members, that is voting and non-voting members, respectively; and
 - (e) must set out-
 - (i) the qualifications for membership;
 - (ii) the process for applying for membership;
 - (iii) any initial or periodic cost of membership in any class;

- (iv) the rights and obligations, if any, of membership in any class; and
- (v) the grounds on which membership may, or will, be suspended or lost.

5 Directors

- (1) If a non-profit company has members, the Memorandum of Incorporation must-
 - (a) set out the basis on which the members choose the directors of the company; and
 - (b) if any directors are to be elected by the voting members, provide for the election each year of at least one-third of those elected directors.
- (2) If a non-profit company has no members, the Memorandum of Incorporation must set out the basis on which directors are to be appointed by its board, or other persons.
- (3) A non-profit company must not provide a loan to, secure a debt or obligation of, or otherwise provide direct or indirect financial assistance to, a director of the company or of a related or inter-related company, or to a person related to any such director.
- (4) Sub-item (3) does not prohibit a transaction if it-
 - (a) is in the ordinary course of the company's business and for fair value;
 - (b) constitutes an accountable advance to meet-
 - (i) legal expenses in relation to a matter concerning the company; or
 - (ii) anticipated expenses to be incurred by the person on behalf of the company;
 - (c) is to defray the person's expenses for removal at the company's request; or
 - (d) is in terms of an employee benefit scheme generally available to all employees or a specific class of employees.

RULES OF MONAGHAN FARM HOMEOWNERS ASSOCIATION NPC

1. INTERPRETATION & DEFINITIONS

- 1.1. These Rules shall be read in conjunction with, and form a part of, the Memorandum of Incorporation of the Monaghan Farm Homeowners Association NPC.
- 1.2. The provisions of the MOI governing Interpretation and Definitions shall apply to these Rules.

2. GENERAL

- 2.1. The Board shall have the authority to determine, amend, and impose such penalties or fines in respect of any infringement of the MOI, these Rules, the Regulations, Architectural rules, Building rules, and house rules as may from time-to-time be in force upon Monaghan Farm.
- 2.2. Members shall at all times be held responsible for the conduct of those residing upon their Erf, their employees, service providers, and visitors (together, "guests") while those persons are on Monaghan Farm.
- 2.3. Members and their guests shall be further bound by the applicable provisions of the Conditions of Establishment and the EMP.
- 2.4. All houses and structures shall be properly maintained and at all times kept in a good and sound state of repair.
- 2.5. Pedestrians and wildlife shall at all times have right-of-way over motor vehicles on the Property.
- 2.6. The maximum permissible speed of any vehicle on Monaghan Farm shall be forty kilometers per hour (40 kph) and the Board may, from time to time, determine -
 - 2.6.1. a lesser speed limit applicable to Monaghan Farm in part or in whole;
 - 2.6.2. other provisions governing the access to Monaghan Farm, and conduct thereon, of vehicles and their drivers;
- 2.7. The number of vehicles motorised or non-motorised and which for the purposes of this clause shall include trailers, caravans, boats, and other wheeled devices of comparable dimension permanently or regularly housed on any erf shall not exceed the number of garages or carports erected upon that erf.

- 2.8. Only 'silent' alarm systems connected in the prescribed manner to the security services employed upon Monaghan Farm shall be permitted.
- 2.9. No petrol, diesel, or other combustion powered generators shall be permitted for domestic use.
- 2.10. Domestic animals shall be permitted upon Monaghan Farm save that -
 - 2.10.1. The Board is to be guided at all times by the rules and regulations of the Monaghan Farm Home Owners Association Memorandum of Incorporation and the local municipal by laws.
 - 2.10.2. Residents should be encouraged to resolve nuisance incidents amongst themselves before involvement of management and the Board.
 - 2.10.3. The number of cats and / or dogs permitted shall at all times be limited to two dogs and / or two cats.
 - 2.10.4. Animals other than cats and dogs may be considered as follows;
 - 2.10.4.1. Pigs and roosters / cockerels / peacocks / geese / ducks / pigeons/ outdoor birds /aviaries;
 - 2.10.4.2. The Board shall **not** permit the keeping of any of these types of animals / poultry.
 - 2.10.5 Rabbits and chickens: on successful application, and provided they are registered, sterilised and contained in an enclosure approved by MAC, residents shall be entitled to keep no more than two rabbits and four chickens. In the event that any nuisance (including smell, noise and agitation to other animals) is caused then management shall be entitled to revoke any permission granted and the resident shall be required to remove the animals at his/her/its own cost.
 - 2.10.6 All cats and dogs are to be registered by their owners on the portal and information on the portal is to be kept up to date by the owners.
 - 2.10.7 All permitted animals shall wear an identification device / collar at all times and all cats are to be sterilised. Proof of neutering is to be uploaded to the portal by the owner.
 - 2.10.8 All animals are to be contained securely at all times. In the event that a home has been designed in such a way that containment not viable or possible, then such owner

shall not be allowed to keep animals. Security and owners shall be encouraged to report all incidences of roaming animals to management immediately. Owners are cautioned that the electric collar system of containment is not always reliable and the onus remains on the owner to ensure the animal/s is/are securely contained at all times.

- 2.10.9 The handler of a dog will be responsible for the supervision and control of the behavior of that dog. When outside the boundaries of their own property, but within the Estate, the handler shall keep the dog on a leash at all times save for within the boundary of any demarcated off-lead parks / areas. Management will be entitled to deal with any unaccompanied dog found roaming freely in a manner that they deem fit.
- 2.10.10 Owners/tenants are encouraged to remove any animal faeces from common / improved property including the roads.
- 2.10.11 The following breeds of dogs shall not be permitted on Monaghan Farm;
 - 2.10.11.1 Pit bulls
 - 2.10.11.2 Shar-pei
 - 2.10.11.3 The Board reserves the right to add to this list from time-to-time.
- 2.10.12 Owners/ Tenants leaving the Estate for extended periods are encouraged to send their dogs to a kennel to reduce noise and barking nuisance.
- 2.10.13 In the event of management being unable to remedy reported nuisance from barking then in such event management shall be entitled to require that the offending animal/s be kept inside and / or to remove the animals to a local kennel at the expense of the owner.
- 2.10.14 All wild animals, including but not limited to leguaan, porcupine scrub hare, mongoose, hedgehog, dyker, jackal and birds encountered on/in the Estate must be considered to be protected species and not unreasonably interfered with. The presence of snakes and beehives must be reported to management.
- 2.10.15 No owner/ tenant shall be permitted to breed any pets / animals.
- 2.11. Nowhere upon Monaghan Farm, without the express proir approval of the Board
 - 2.11.1. shall any firearm, air gun, pistol, catapult, bow, or any other weapon be used or discharged; nor

2.11.2. shall any firework be used.

3. ARCHITECTURE

- 3.1. The Monaghan Aesthetics Committee ("MAC") shall be responsible for maintaining the overall aesthetics of the development and assessing the site development, building, and landscaping plans submitted for approval prior to construction against the architectural guidelines and landscaping guidelines from time-to-time in force as well as monitoring compliance with approved plans during and upon completion of construction.
- 3.2. No architect may be appointed in respect of any works unless and until they have been approved for work upon Monaghan Farm by MAC.
- 3.3. No principal building contractor may be appointed unless and until they have been approved for work upon Monaghan Farm by the Board, MAC, or the Manager.
- 3.4. Construction works shall not commence unless and until any conditions to and approval granted by MAC have been fulfilled and unless and until all relevant local authority approvals have been granted.
- 3.5. The construction period in respect of any residential erf under construction whilst the National Lockdown Regulations promulgated in terms of the Disaster Management Act, Act 57 of 2002 are in effect and force, shall not exceed twenty-one (21) months, as calculated from commencement to completion.
- 3.6. The Member shall at all times be responsible for ensuring compliance by their architects and contractors with all the Association's architectural, landscaping, and building requirements.
- 3.7. Rule deleted (by way of Board resolution 9 July 2021).
- 3.8. Minimum standards which shall inform the discretion of MAC in respect of the application of the architectural and landscaping guidelines shall include
 - 3.8.1. application of the requirements of the Conditions of Establishment and the EMP;
 - 3.8.2. consideration of the views and privacy in respect of neighbouring erven;
 - 3.8.3. prohibition upon the erection of any physical barrier on the boundary of an erven save, at the discretion of MAC, where that boundary corresponds also with the boundary of the approved development pocket;

3.9. MAC shall –

- 3.9.1. operate under the control of the Developer until 31 December 2018 whereafter it shall operate under the control of the Association and its Board;
- 3.9.2. from time-to-time set or amend
 - 3.9.2.1. the architectural and landscaping guidelines;
 - 3.9.2.2. the processes to be followed for consideration of plans submitted to it for approval; and
 - 3.9.2.3. the fees to be paid in respect of plans submitted to it for approval.
- 3.10. Subsequent to the completion sign-off by MAC of any works conducted on a development, no alterations or addition shall be permitted and nothing may be further erected whether temporary or permenent, without application having been made to, and approval granted by, MAC and which restriction shall also apply in respect of external play equipment, jungle gyms, statues, garden sheds, and their like.

4. OPEN SPACES

- 4.1. No person shall, nor cause nor permit, upon Monaghan Farm
 - 4.1.1. the harm, disturbance or destruction of any wild animial, insect reptile, or bird;
 - 4.1.2. the disturbance, destruction, or collection of any plant material living or dead without the consent, or upon the direction, of the Board;
 - 4.1.3. the pollution of any watercourse or dam;
 - 4.1.4. the lighting of any open fire save in such place as designated that purpose by the Board;
 - 4.1.5. the disposal of litter, cigarette butts and similar waste save in such receptacles and in such places set aside for the purpose by the Association;
 - 4.1.6. conduct which may interfere with the enjoyment of other Members of the open spaces.
- 4.2. The Board may, from time-to-time, establish and amend provisions governing
 - 4.2.1. access to, and use of, the river, riverfront, and other watercourses and wetlands;

- 4.2.2. camping; and
- 4.2.3. restrictions upon the use of, or access to, specific open spaces.

5. EDUCATION

Pursuant to the local authority approvals pertaining to the development, the Association has assumed certain obligations in respect of education which it shall discharge as follows.

- 5.1. Purpose built premises suitable for the operation of a pre-school have been constructed by the Developer on Erf 181 for transfer to the Association who may then by rent or by sale of those premises to an independent operator cause that education be made available to not more than 100 (one hundred) pupils.
- 5.2. A school has been established on Erf 289 for operation by Curro Holdings which will be progressively expanded to accommodate not more than 1300 (thirteen hundred) pupils and in respect of which the Association has
 - 5.2.1. caused the provision by Curro Holdings of educational busaries on the basis that
 - 5.2.1.1. one bursary representing 100% of the normal school fee payable will be provided for each of up to three classes within each annual Grade R intake such that if three bursaries are provided in respect of thirteen consequitve annual intakes there shall be 39 (thirty nine) bursaries simulataneously provided across those thirteen grades, and
 - 5.2.1.2. each bursary shall, in respect of each receipient, endure through to Grade 12 matriculation subject to the attaiment of minimum standards for conduct and academic acheivement; and
 - 5.2.1.3. each bursary shall be awarded to under-priveleged children as determined by aptitude and means testing with preference given to qualifying pupils of the Cradle Pre-School.
 - 5.2.2. undertaken to support the beneficiaries of those bursaries in respect of the requisite uniforms and educational materials; and
 - 5.2.3. undertaken to make available Erf 290 for an annual rental of R100 (one hundred rands) to accommodate sports fields and other sporting facilities established at the cost of the school operator.

5.3.	The Association has further taken responsibility for the operating costs of a pre-school facility operated in the Cradle Nature Reserve, or in a comparable facility, for not more than 40 (forty) pupils.

REGULATIONS OF MONAGHAN FARM HOMEOWNERS ASSOCIATION NPC

1. INTERPRETATION & DEFINITIONS

- 1.1. These Regulations shall be read in conjunction with the Memorandum of Incorporation of the Monaghan Farm Homeowners Association NPC and its attendant Rules.
- 1.2. The provisions of the MOI governing Interpretation and Definitions shall apply to these Rules.

2. ARCHITECTURE, BUILDING, & PLANS

- 2.1. Development, building, and landscaping plans must be informed by, and be consistent with,
 - 2.1.1. the Conditions of Establishment and the EMP;
 - 2.1.2. the Rules attaching to the MOI; and
 - 2.1.3. the published prevailing rules and/or guidelines in respect of architecture, building, and landscaping.
- 2.2. Plans must be submitted to MAC for its approval, together with a R5,000 submission fee whereupon, and after due consideration, MAC shall either
 - 2.2.1. grant its unconditional approval to the plans by endorsing each of the approved plans and returning them to the architect, or
 - 2.2.2. make written comment to, or upon, each of the plans and return them to the architect for amendment and resubmission.
- 2.3. MAC shall require that
 - 2.3.1. all exposed plumbing and washing lines shall be fully screened from street elevation and any other elevation from which it may be visible;
 - 2.3.2. all communication instruments such as telephones, fax, and internet must be connected to the fibre-optic network made available at the boundary of each erf and no independent, overhead, or otherwise exposed connection shall be permitted.
- 2.4. During the plan approval process, MAC shall communicate solely with the architect acting as agent for the owner.

- 2.5. Only when approvals are granted by MAC without condition may application for local authority approval be made.
- 2.6. Pending receipt of plan approval from the local authority, MAC may permit limited site clearance works to commence.
- 2.7. No encroachment over the boundaries of the erf shall be permitted during construction.

3. TRAFFIC & VEHICLES

- 3.1. Except in respect of a golf cart, no person shall drive any vehicle upon Monaghan Farm if
 - 3.1.1. not holding a valid personal license which would permit the driving of such vehicle upon the public roads of the Republic of South Africa; or
 - 3.1.2. not carrying a valid license or permit specific to that vehicle as would be required for its use upon the public roads of the Republic of South Africa; or
 - 3.1.3. (Regulation deleted and replaced by a new Regulation 3.5 by way of special resolution 17 September 2022)
- 3.2. No vehicle shall, without the prior approval of the Manager
 - 3.2.1. leave the improved road network of Monaghan Farm;
 - 3.2.2. enter or exit Monaghan Farm at any point other than the main entrance or contractors gate;
 - 3.2.3. fail, on entry or exit, to stop at the relevant security checkpoint;
 - 3.2.4. enter or exit without being cleared at the relevant security checkpoint; or
 - 3.2.5. be parked or left unattended in a common area for more than 7 (seven) days following which it will be removed at the owners risk and expense.
- 3.3. Vehicles motorised or non-motorised and which for the purposes of these regulations shall include trailers, caravans, boats, and other wheeled devices of comparable dimension
 - 3.3.1. shall not be parked or stored outside the development pocket without the prior approval of the Manager and shall at all times, and
 - 3.3.2. shall, within the development pocket, be parked or stored out of sight from the neighbouring streets and homes.

- 3.4. All damage to Monaghan Farm caused by any vehicle under the control or command of a Member, those residing upon their Erf, their employees, service providers, and visitors (together, "guests") shall be the responsibly and for the account of, the Member.
- 3.5. Two-stroke motorbikes may only be operated on Monaghan Farm under the following conditions:
 - 3.5.1. as a means of transport to get on and off the Monaghan Farm traversing between the main gate and the resident's home; and
 - 3.5.2. only on the tar roads only using the shortest route between the resident's home and the main gate and vice versa; and
 - 3.5.3. no performing of any wheelies, stunts, and / or any unusual use of the motorbikes are allowed; and
 - 3.5.4. no revving of motorbikes is allowed; and
 - 3.5.5. no guests are allowed to ride onto the Monaghan Farm with their two-stroke motorbikes. Guests are required to wait for the resident concerned to meet them at the main gate; and
 - 3.5.6. the usage times for traversing on two-stroke motorbikes on Monaghan Farm may be restricted on weekends and public holidays in the sole and absolute discretion of the Board of Monaghan Farm NPC; and
 - 3.5.7. the operation of two-stroke motorbikes is subject to Rule 2.6 and Regulation 3."

4. ENVIRONMENTAL

- 4.1. A designated environmental consultant shall inspect Monaghan Farm on an annual basis to ensure that proper environmental management and the terms of the EMP are upheld.
- 4.2. Save as otherwise permitted or provided for by the Association or the Manager as the case may be; no person shall
 - 4.2.1. launch upon the river, watercourse, or dam any craft of any description;
 - 4.2.2. fish in the river or any dam without a catch-and-release permit issued by the Manager.

5. DOMESTIC REFUSE

- 5.1. The removal of domestic refuse shall be under the control of the Manager who may
 - 5.1.1. determine the number, type, and size of refuse containers to be utilised upon each erf;
 - 5.1.2. make direction in regard to the placement of refuse containers for the purposes of refuse collection;
 - 5.1.3. require payment on a reasonable basis for the provision of refuse containers;
 - 5.1.4. determine the manner in which refuse shall be separated prior to collection from the erf for the purposes of recycling.

5.2. No refuse shall -

- 5.2.1. be held on an erf or presented for collection except within the prescribed refuse containers; or
- 5.2.2. placed outside the erf except in the manner and place which the Manager may from time-to-time determine.

6. DOMESTIC ANIMALS

- 6.1. The Board may, in writing, and at its absolute discretion -
 - 6.1.1. permit the keeping of more than two dogs or two cats on any erf;
 - 6.1.2. permit the keeping of any animal other than a dog or cat;
 - 6.1.3. determine parts of the common areas where dogs may be permitted off-leash while under the supervision of their owner;
 - 6.1.4. authorise the Manager to
 - 6.1.4.1. impound any animal found wandering Monaghan Farm unsupervised; and
 - 6.1.4.2. impose a fine of R1,000 or greater in respect of each instance a dog is found wandering Monaghan Farm unsupervised or off-leash outside an area that may have been designated for that purpose
 - 6.1.5. determine that animals which are found to be a nuisance to other residents shall be required to be

- 6.1.5.1. kept inside at night; or
- 6.1.5.2. removed from Monaghan Farm.

7. EQUESTRIAN

- 7.1. The Developer shall build for the use by, and at some future time transfer ownership to, the Association the equestrian facilities consisting
 - 7.1.1. stables for 30 (thirty) horses;
 - 7.1.2. outside paddocks for 15 (fifteen non-stabled horses)
 - 7.1.3. a lunging ring;
 - 7.1.4. two arenas;

7.2. The Board may determine

- 7.2.1. to operate the equestrian facilities in the name and for the account of the Association;
- 7.2.2. rent the equestrian facilities to a third-party operator;
- 7.2.3. construct up to 15 (fifteen) additional stables at the cost of the Association to satisfy demand
- 7.2.4. all subject to the limitation that not more than a total of sixty (60) horses may be housed on Monaghan Farm at any time.

8. FINANCIAL

- 8.1. Any penalty imposed by the Manager, the Board, or the Association shall be applied to, and form part of, the Member's levy.
- 8.2. The Association shall be entitled to charge interest on all overdue amounts at a rate equal to the prime rate of interest charged from time-to-time by Nedbank plus 5% (five percent).
- 8.3. All levies are due and payable not later than the seventh day of each month.
- 8.4. No erf may be transferred without a clearance certificate in respect of all monies owing issued by the Association and valid until at least the end of the month of transfer.

9. GENERAL

- 9.1. All persons on Monaghan Farm shall adhere to all regulations or house rules issued in respect of security and Members and their guests shall at all times give their co-operation to, and comply with the requirements of, the security personnel.
- 9.2. (Regulation 9.2 deleted, by way of special resolution 17 September 2022)
- 9.3. Sports facilites shall be for the use of Members and their guests except that the Board may
 - 9.3.1. permit their use from time-to-time by other persons or organisations;
 - 9.3.2. implement a system of booking and reservation where justified by demand.
- 9.4. In respect of an ongoing infringement of the Rules or Regulations of Monaghan Farm NPC , the penalty specified in the Schedule of Penalties may be imposed until such time as the infringement is rectified. A monthly penalty is payable per calendar month and if an infringement stops part way through a calendar month, then the monthly penalty is payable on a pro-rata basis.
 - 9.4.1. Should there be a persistent disregard for the penalties imposed and the infringement is not rectified within a minimum of 3 months, the Board of Monaghan Farm NPC reserves the right to approach the Community Schemes Ombud Service for legal intervention and/or pursue other legal avenues to remedy the infringement.
 - 9.4.2. Depending on the seriousness of the infringement and the potential danger it entails to other Members and / or Homeowners, the Board of Monaghan Farm NPC may deviate from the minimum 3-month period within its sole and absolute discretion.
 - 9.4.3. The imposition of a penalty does not exclude a Homeowner's liability to Monaghan Farm NPC or to any other person in respect of damage to property, injury, or death to persons."

9A . Noise reduction

9A.1 "Noise nuisance" means any sound which disturbs or impairs or may disturb or impair the convenience or peace of any person within the estate.

- 9A.2 Excessive noise shall be described as exceeding what is proper, necessary or normal.
- 9A.3 All residents should respect the rights of other residents by not allowing or causing excessive noise.
- 9A.4 No noise shall be permitted during quiet hours, which may be imposed by the Board of Monaghan Farm NPC in their sole and absolute discretion in order to minimise any nuisance.
- 9A.5 Without limiting the generality of the aforegoing
 - 9A.5.1. Music systems, video players, DVD players, televisions and musical instruments, and the like must be played and operated at a noise level so as not to cause a disturbance to other residents.
- 9A.6 Recreational vehicles as contemplated in Regulation 3 may not be driven or operated on a residential erf or Monaghan Farm in a manner that causes a nuisance to any other resident.
- 9A.7 No activity or hobby, which causes aggravation, nuisance or noise pollution to other owners, may be exercised on Monaghan Farm.
- 9A.8 The revving of any engines and motorbikes is not permitted.
- 9A.9 The Board may impose restrictions upon the use of certain equipment generally or at weekends and upon public holidays in order to minimise the potential nuisance to neighbours in respect of-
 - 9A.9.1. the use of power tools, powered lawnmowers, brush cutters, weed eaters and similar machinery is only permitted on the following days during the following hours:
 - 9A.9.1.1 Monday to Friday from 08H00 to 16H00, unless these days fall on a public holiday; and Saturday from 08H00 to 14H00.
 - 9A.9.2. the use of power tools, powered lawnmowers, brush cutters, weed eaters and similar machinery is prohibited on Sundays and Public Holidays.
- 9A.10 Should any dispute arise concerning the measuring of excessive noise levels, the Gauteng Noise Control Regulations as set out in Government Gazette 5479 of 1999 as amended from time to time may be used by the Board of Monaghan Farm NPC.

The Board, or its representative's decision as to whether there is a breach of these Regulations shall be final."

9B. Use of generators for construction purposes

- 9B.1 The use of generators may only be used for construction purposes and only on new building sites.
- 9B.2 The registered owner, or his appointed contractor, must apply for an electricity connection and account with Eskom within at least 90 days of breaking ground.
- 9B.3 During the construction period, and only if there is an electricity interruption such as loadshedding, may a generator be used on construction sites.
- 9B.4 A generator may only be operated subject to the above, during weekdays between 06H30 and 18H00.
- 9B.5 Should the electricity connection with Eskom be interrupted for some other reason than loadshedding, the registered owner or his appointed contractor must apply to the Monaghan Farm NPC Management for the use of a generator in the interim on good cause shown.
- 9B.6 The Management of Monaghan Farm NPC reserves the right to retract the permission for generator use should a generator cause a disturbance."



Revision 17 September 2022

In terms of paragraph 11.9 of the Memorandum of Incorporation of Monaghan Farm, the Board of Directors is empowered to levy additional contributions upon specific members for services provided. In terms of paragraph 16.1.3 of the Memorandum of Incorporation of Monaghan Farm, the Board of Directors is empowered to impose a system of penalties for the enforcement of any of the Estate Rules, the amounts of penalties to be stipulated in the Regulations determined by the Board of Directors. In respect of these powers, herewith the Monaghan Farm Fees, Charges and Penalty Structure.

	IISTRATION: FEES, RENTAL AND SERVICE CHARGES	Fee amount or Penalty amount
-	Aesthetics Committee plan submission,	
	adjudication, and inspection fees	
	**All Aesthetics Committee fees are to be paid	
	directly to the service provider. **	
Ple	ase request banking details and booking reference	
via	mac@monaghanfarm.co.za .	
a)	Home inspections: as a requirement to issue a levy clearance certificate.	R 1 000.00 upfront payment, per inspection.
b)	Landscape plan submission.	R 575.00 upfront payment, per submission.
c)	Renovations: Small Additions Non council submissions e.g., Fencing, play area equipment, braai area, driveways, PV panels/solar, aircon/evaporative cooling, cctv, lighting(exterior), other/ all external elements not shown on MAC approved plans.	R 500.00 upfront payment, per submission.
d)	Renovations: Additions max 100m² Requiring submission to council / regulatory authority. e.g., Pools, Carports, additions & alterations (enclosed/roofed areas)	**Budget around R 8 500.00 for the total submission process. Please pay per individual submission only.**
	Submission 1 - concept Submission 2 - concept/ design development Submission 3 - council Submission 4 - amendment landscape plan MAC Sign off / final inspection.	R 3 000 .00 upfront payment, per submission. R 1 500 .00 upfront payment, per submission. R 1 500 .00 upfront payment, per submission. R 1 000 .00 upfront payment, per submission. R 1 500 .00 upfront payment, per submission.
	Additional meetings, 30 min charged at	R 1500 .00 upfront payment, per additional submission.



 a) Renovations exceeding 100m² & new build projects. Requiring submission to council / regulatory authority. 	**Budget around R 12 500.00 for the total submission process. Please pay per individual submission only.**
Submission 1 - concept Submission 2 - concept/ design development Submission 3 - council Submission 4 - Landscape plan - concept Submission 5 - landscape plan approval Submission 6 - approval of materials Submission 7 - variation orders MAC Sign off / final inspection. Additional meetings, 30 min charged at	R 3 000 .00 upfront payment, per submission. R 1 500 .00 upfront payment, per submission. R 1 500 .00 upfront payment, per submission. R 1 000 .00 upfront payment, per submission. R 1 000 .00 upfront payment, per submission. R 1 500 .00 upfront payment, per submission. R 1500 .00 upfront payment, per additional submission.
2) Access cards Annual renewal required. Please note: all access cards will expire month end February of each year.	
a) Residents access cards (where biometrics have failed)	R 500 per card / lost card replacement
b) Cyclist card (granting access for cyclists via the emergency exit gate situated: The Other Side)	R 218 per card / lost card replacement
c) Domestic staff (access cards)	R 218 per card / lost card replacement
3) Administration Fees a) Late payment fee automatically levied where levies are not paid in full and funds do not reflect in the company's bank account by close of business on the 7 th of each month.	R350 per month.
b) Reminder notice fee Per notice issued in accordance with the MFHOA debtors' policy	R300 per month
 Failure to subscribe to the direct debit payment method to settle monthly levies and sundry charges as applicable. 	R300 per month
4) Building / Contractors Deposit (refundable)	



		Revision 17 September 2022
a)	New Build + projects exceeding 150m ²	R 20,000
b)	All other contract / building works	R 10,000
5) Bui	ilders Yard Rentals	
(availa	able to building contractors only)	
a)	With office building	R 6,348 per month
-	Without office building	R 4,021 per month
•	· ·	
6) Gei	neral: Sales & Hire	
a)	Collection of goods	R 158 per delivery
-	Delivery of goods	R 158 per delivery
c)		R 22 per bale
d)		R 54 per bale
e)		R 380
f)	Water meters	At cost, please enquire.
g)	Water meter 'key'	R 110
h)	•	At cost, please enquire.
i)	Refuse bin: Green	At cost, please enquire.
j)	Recycling refuse bags (pack of 20 bags)	R 50
k)		R 380 min fee
,	time _ call out fee.	
	_	
7) Gra	ass cutting and garden refuse collection.	
a)	Brush cutting	Min service charge of R353 calculated at
	(cutting of veld / berms to min 150 mm)	R53 per staff member per hour
b)	Rake, collection, and removal of veld grass	Min service charge of R353 calculated at
	cuttings	R53 per staff member per hour
c)	Garden refuse collection	Min service charge of R353 calculated at
		R53 per staff member per hour
d)	Garden refuse placed in bags and placed for	Collected with household waste at no charge.
	collection on refuse collection days.	
e)	Tractor& tractor equipment hire.	Min charge of R353, thereafter R353 per hour.
	y clearance certificate	R 1 360 per levy clearance certificate issued.



b) A3 10) Security Alarms a) Monthly alarm monitoring b) Home alarm transmitter enabling connection with security control room. PENALTIES: DOMESTIC ANIMALS 11) The keeping of an unauthorised animal, as defined by the pet rules. 12) Contravention of pet rules R2 13) Dogs found roaming the estate. R1	2.20 per page 3.30 per page lo charge at cost, please enquire.
Black & white a) A4 b) A3 10) Security Alarms a) Monthly alarm monitoring b) Home alarm transmitter enabling connection with security control room. PENALTIES: DOMESTIC ANIMALS 11) The keeping of an unauthorised animal, as defined by the pet rules. 12) Contravention of pet rules R1 13) Dogs found roaming the estate. R1	3.30 per page lo charge
a) A4 b) A3 10) Security Alarms a) Monthly alarm monitoring b) Home alarm transmitter enabling connection with security control room. PENALTIES: DOMESTIC ANIMALS 11) The keeping of an unauthorised animal, as defined by the pet rules. 12) Contravention of pet rules R1 13) Dogs found roaming the estate. R1	3.30 per page lo charge
b) A3 10) Security Alarms a) Monthly alarm monitoring b) Home alarm transmitter enabling connection with security control room. PENALTIES: DOMESTIC ANIMALS 11) The keeping of an unauthorised animal, as defined by the pet rules. 12) Contravention of pet rules R1 R2 R3 R3 R6 R6 R6 R7 R1 R1 R1 R1 R1 R1 R1	3.30 per page lo charge
10) Security Alarms a) Monthly alarm monitoring b) Home alarm transmitter enabling connection with security control room. PENALTIES: DOMESTIC ANIMALS 11) The keeping of an unauthorised animal, as defined by the pet rules. pe un 12) Contravention of pet rules R2 13) Dogs found roaming the estate. R1	lo charge
a) Monthly alarm monitoring b) Home alarm transmitter enabling connection with security control room. PENALTIES: DOMESTIC ANIMALS 11) The keeping of an unauthorised animal, as defined by the pet rules. 12) Contravention of pet rules R2 13) Dogs found roaming the estate. R1	•
a) Monthly alarm monitoring b) Home alarm transmitter enabling connection with security control room. PENALTIES: DOMESTIC ANIMALS 11) The keeping of an unauthorised animal, as defined by the pet rules. 12) Contravention of pet rules R2 13) Dogs found roaming the estate. R1	•
b) Home alarm transmitter enabling connection with security control room. PENALTIES: DOMESTIC ANIMALS 11) The keeping of an unauthorised animal, as defined by the pet rules. 12) Contravention of pet rules R2 13) Dogs found roaming the estate. R1	•
b) Home alarm transmitter enabling connection with security control room. PENALTIES: DOMESTIC ANIMALS 11) The keeping of an unauthorised animal, as defined by the pet rules. 12) Contravention of pet rules R2 13) Dogs found roaming the estate. R1	•
PENALTIES: DOMESTIC ANIMALS 11) The keeping of an unauthorised animal, as defined by the pet rules. 12) Contravention of pet rules R2 13) Dogs found roaming the estate.	
PENALTIES: DOMESTIC ANIMALS 11) The keeping of an unauthorised animal, as defined by the pet rules. 12) Contravention of pet rules R2 R3) Dogs found roaming the estate.	
11) The keeping of an unauthorised animal, as defined by the pet rules. 12) Contravention of pet rules R2 R3 R3 R1 R1	
11) The keeping of an unauthorised animal, as defined by the pet rules. 12) Contravention of pet rules R2 R3 R3 R1 R1	
by the pet rules. 12) Contravention of pet rules R2 13) Dogs found roaming the estate. R1	
12) Contravention of pet rules R2 13) Dogs found roaming the estate. R1	1 000 instant penalty per animal identified. R1 000
12) Contravention of pet rules R2 13) Dogs found roaming the estate. R1	er month per unauthorised animal thereafter until
13) Dogs found roaming the estate.	nauthorised animal is removed.
13) Dogs found roaming the estate.	
13) Dogs found roaming the estate.	2 000 per contravention.
	1 000 per dog and R1 000 per transgression
14) Dogs off lead R1	1 000 per dog and K1 000 per transgression
14) Dogs on lead NI	1 000 per dog and R1 000 per transgression
	11 000 per dog and K1 000 per transgression
DENIALTIES, TRAFFIC AND DARVING	
PENALTIES: TRAFFIC AND PARKING	
15) Not using designated parking area's or R5	500 per vehicle per transgression.
16) Parking on grass or grass swales R5	500 per vehicle per transgression.
17) Failing to adhere to traffic signs (e.g., failure to R5	500 per transgression
stop at stop street etc)	
18) Contravening the contractual arrangement how to	
use internal roads - prescribed speed limit of 40 km	
per hour by: -	
· · · · · · · · · · · · · · · · · · ·	550 per contravention
•	220 per contravention
·	1 000 non contraction
·	1 000 per contravention
· · · · · · · · · · · · · · · · · · ·	1 500 per contravention
e) Repeat of any traffic offence within a 90-day	·
period will result in a 100% escalation of the	1 500 per contravention
penalty amount.	1 500 per contravention
d) +30 km per hr	·



PENALTIES: HOUSKEEPING		
TRANSGRESSION	PENALTY Raised where a transgression notice has been issued and corrective action is not completed within a specified timeframe.	
19) Walls, window, and door frames – poorly maintained	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.	
20) Wall & timber paint/ timber stain fading /flaking and warping	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.	
21) Satellite disha) that is not painted the same color upon the surface / wall it is installed.	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.	
b) unapproved installation /position	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.	
22) Exposed wires/ pipes	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.	
23) Berms – that are not establishment & planted	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.	
24) Berms – No Dust suppression - e.g., hessian / irrigation	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.	



	Nevision 17 September 2022
25) Fences - poorly maintained	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
26) Gutters & roof sheeting lifting & buckling	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
27) Roof sheeting, gutters, and water tanks fading/ peeling	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
28) Awnings/ sunscreens/ pergolas – poorly maintained	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
29) Incorrect wall & window frame colour	R2 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R2 000 monthly thereafter until the breach is remedied.
30) Planting of unapproved trees and shrubs	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied –including an obligation to submit in terms of the Architectural Rules a New Landscape Plan
31) Unapproved Compost heaps (placement)	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
32) Unapproved Sheds	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
33) Unapproved Jungle gyms	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out



	the breach and R1 000 monthly thereafter until the breach is remedied.
34) Unapproved Fences/ temp. fencing & Gates	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
35) Unapproved External lighting	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
36) Unapproved Pot plants/ statues/ structures	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
37) Unapproved signage/house nos. (Only HOA installed signs permitted)	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
38) Unapproved awnings	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
39) Exposed heat pumps, pumps not screened from view	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
40) Unapproved instillation of evaporated Cooling Units, Heat Pumps, Solar Panels, and the like	R2 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R2 000 monthly thereafter until the breach is remedied.
41) Unapproved Chicken coops	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out the breach and R1 000 monthly thereafter until the breach is remedied.
42) Unapproved Chimney cowls	R1 000 per month 60 Days after transmittal of a written demand in terms of Paragraph 16.1.1 of the Memorandum of Incorporation setting out



	Revision 17 September 2022
	the breach and R1 000 monthly thereafter until
	the breach is remedied.
	R1 000 per month as well as monthly thereafter
	60 Days after transmittal of a written demand in
43) Unapproved Vegetable gardens	terms of Paragraph 16.1.1 of the Memorandum
	of Incorporation setting out the breach until the
	breach is remedied.
	R1 000 per month 60 Days after transmittal of a
	written demand in terms of Paragraph 16.1.1 of
44) Exposed washing line	the Memorandum of Incorporation setting out
,,	the breach and R1 000 monthly thereafter until
	the breach is remedied.
	R1 000 per month 60 Days after transmittal of a
	written demand in terms of Paragraph 16.1.1 of
45) Exposed refuse bins	the Memorandum of Incorporation setting out
, ב	the breach and R1 000 monthly thereafter until
	the breach is remedied.
	R1 000 per month 60 Days after transmittal of a
	written demand in terms of Paragraph 16.1.1 of
46) Incorrectly stored / exposed garden equipment	the Memorandum of Incorporation setting out
40) moorrectly stored / exposed garden equipment	the breach and R1 000 monthly thereafter until
	the breach is remedied.
	R1 000 per month 60 Days after transmittal of a
	written demand in terms of Paragraph 16.1.1 of
47) Visible temporary pools and or trampolines	the Memorandum of Incorporation setting out
47) Visible temporary pools and or trampolines	the breach and R1 000 monthly thereafter until
	the breach is remedied.
	R1 000 per month 60 Days after transmittal of a
	written demand in terms of Paragraph 16.1.1 of
48) Exposed garages & clutter	the Memorandum of Incorporation setting out
46) Exposed garages & clutter	•
	the breach and R1 000 monthly thereafter until
	the breach is remedied.
	R1 000 7 Days after transmittal of a written
49) Exposed caravans & trailers (please make use of a	demand in terms of Paragraph 16.1.1 of the
trailer park)	Memorandum of Incorporation setting out the
trailer park)	breach. Then R2000 monthly thereafter until the
	breach is remedied.
	R1 000 7 Days after transmittal of a written
	demand in terms of Paragraph 16.1.1 of the
50) Unsightly garden waste	Memorandum of Incorporation setting out the
	breach. Then R1000 monthly thereafter until the
	breach is remedied.



HOUSE RENOVATIONS & BUILDING CONTRACTORS PENALT	Y STRUCTURE
TRANSGRESSION	PENALTY CHARGES WITH EFFECT JAN 2021
51) Spillage of hydrocarbons	R2 500 per incident & rehab costs
52) Drip trays not being used under vehicles, trucks, TLB,	R1 500
generators and any hydrocarbon machinery.	
53) Indigenous plants of any description may not be moved	R10 000 (Tree)
from the stand without approval of the ECO. (NEW	R 1 000 (Other)
BUILDS ONLY)	
54) Construction rubble shall always be placed in a Skip and	R2 000(No container)
covered. Skips must be emptied on a regular basis.	R 500 (Not covered)
	R1 000 (Not being emptied)
55) Encroaching onto adjacent stands or works exceeding	R2 000
50% of the development pocket of the site.	
56) Not using appropriate trays for cleaning equipment	R1000 per incident
e.g., paint brushes, Rhyno Lite and the like.	
57) No concrete, cement, or other building materials to be	
stored, mixed, or prepared on roadways, kerbs,	R2 000 per incident
pavements or outside of the 50% development pocket.	
pavements of outside of the 50% development position	
Spillage of cement or concrete on the roadways.	R 3000 per incident & Rehab/ clean-up costs
58) Non - compliance of Health & Safety regulations in any	R2 000 per incident
manner identified.	'
NB: Following Non-Compliance with any Health and	
Safety Regulations, work will need to cease	
immediately on written demand by Monaghan Farm in	
terms of its entitlement in terms of paragraph 16.1.2 of	
its Memorandum of Incorporation. The building site will	
remain closed until breach(es) is/are remedied to the	
satisfaction of Monaghan Farm and any Statutory	
Investigations and/or Inspections are completed.	
59) Trucks –	
Exceeding the below restriction in any manner:	
a) 6m² tipper trucks only permitted.	R10 000 per transgression identified
b) Fixed chassis vehicle only	
 c) No articulated machinery permitted. 	
d) Brick loads limited must be limited to 6000	
bricks.	
60) Failure to submit to security search/security protocol	ACCESS TO SITE OR MONAGHAN FARM ESTATE
	REVOKED.
61) Development pocket to be screened off on all sides	
with 80% density black shade cloth, at 1,4m height, for	
the duration of building operations.	
Not using appropriate:	
a) Y stands,	



	Revision 17 September 2022
	1 000 per week of the offence until the breach is
c) 3 strand wire.	remedied.
Eailure to keep site hearding and fencing in good order	2000 per offence and thereafter
,	2000 per offence and thereafter 500 per day until the breach is remedied.
not property reinstated	300 per day dritti the breach is remedied.
not property remstated	
Failure maintain and ensure a lockable gate is provided R	2000 per offence and thereafter
•	500 per day until the breach is remedied.
	. ,
NB: Following Non-Compliance work will need to	
cease on written demand as Monaghan Farm is	
entitled in terms of paragraph 16.1.2 of its	
Memorandum of Incorporation, and site will be closed	
until breach(es) is/are remedied to the satisfaction of	
Monaghan Farm.	
1	10 000 per incident
yard or causing a veld fire on site or within the estate.	10 000
·	10 000 per incident together with an additional
the Builder's Yard on foot.	R500 per person as an additional penalty
64) All Vehicles are to be parked within the development	charge
pocket of a Stand / Erf.	
pocket of a standy Lin.	
Vehicles parked on tar roads require written permission	
from the estate manager.	
, and the second	
Vehicles not correctly parked (per above)	1 000 penalty per vehicle.
, ,	1 000 per incident
development pocket or existing roads.	
Description of the state of the	
Permission may be obtained from the estate manager.	
66) Undertaking construction outside of the allowed times. R2	2 500 per hour
(6h30-18h00)	2 300 per 110ui
	1 000 or such amount of penalty as the Board of
regulations not mentioned above.	Monaghan Farm may resolve upon in their sole
<u> </u>	and absolute discretion
68) Damage to Curbs.	2000.00 plus cost of repair or
1	eplacement.
69) Damage to Swales.	2000.00 plus cost of repair or
re	placement.
	praceriierie.
	2000.00 plus cost of repair or



	•
71) Encroaching onto road reserve or servitudes.	R5 000
72) Damage to any Township Services or services	R3 000 plus cost of repair or
infrastructure such as, but not limited to water, water	replacement.
meters, sewer, electrical or telecoms and the like.	
73) Failure to report to Monaghan Farm Management any	R5 000 plus cost as per 68 – 72 of this schedule.
damage caused, whether wilful or not.	
74) Tampering with any of the Township Services or	R5 000 per incident
services, including, but not limited to forcing open	
water meters.	
75) Principal contractors leaving site before subcontractors.	R10 000 per incident
76) Overloading vehicles with building materials.	R1 500 per incident
77) Overloading vehicles with staff,	R1 500 per incident
(e.g Hanging out of open tailgate.)	
78) Littering from stationery or moving vehicles.	R1 500 per incident
79) Not using a portable chemical toilet on site or failure to	R2 000 per incident
provide sufficient toilets as prescribed	
80) Please ensure that all variation orders, AS-Builds etc	R15 000 per incident
are approved by MAC before work can commenced.	
NB: Any unapproved construction will be liable for a	
penalty and may risk having to be demolish.	
81) (a) Smoking of Cigarettes within 10 meters of the	R500 per incident
Contractors Airlock gate.	
82) Lack of builder's yard & development pocket	R750 per week until remedied.
maintenance/housekeeping.	
83) Builders yard containers/huts within the builders yard	R750 per week until remedied.
not painted grey.	
84) Any Expired Internal or Regulatory Documentation for	R1000 per week until remedied
example – expired letter of good standing / Other	
regulatory certification that may have expired etc.	
(Site shall be closed after 30 days on written demand	
in terms of Monaghan Farm's entitlement in terms of	
paragraph 16.1.2 of its Memorandum of	
Incorporation)	
85) Disposal of building material at a non-authorized refuse	R5 000 per incident.
site.	
86) Not erecting builder's board within 14 days of breaking	R500 per day until remedied
ground.	
The site shall be closed on day 21 on written demand	
in terms of Monaghan Farm's entitlement in terms of	
paragraph 16.1.2 of its Memorandum of Incorporation	
until rectified.	
87) Any vehicle for delivery/contract work found driving	R2 000 per incident.
	NZ 000 per meident.
around the estate canvassing work opportunities	NZ 000 per meident.
88) On repeat offence-Where the CEO/Estate Manager has pointed out a contravention of the rules and the	N2 000 per meident.



Revision 17 September 2022

contractor has not remedied the situation within 48 hours a 100% an escalation of the penalty will be applied, and the Contractor may be locked out of Monaghan Farm Estate on written notification in terms	
of Monaghan Farm's entitlement in terms of paragraph 16.1.2 of its Memorandum of Incorporation.	
89) Cement trucks found dropping off cement along the outside of Monaghan Farm.	R7 500 per incident.
90) Failure to permanently have available on-site Council	R1 000 per incident.
approved plans, MAC approved plans, MAC approved variation orders and approved landscape plan.	
91) Staff/contractors not in uniform.	R1 000 per person.
92) Connecting to the HOA sewer system prior to a flush inspection	R15 000 per incident.

BUILDING PENALTY

93) Delay in completion of Building works applicable till 30 September 2023	Failure to complete a building within 21 months
	from date of commencement of building works,
	will accrue a monthly building penalty levy equal to
	a double monthly standard levy up to and including
	36 months from date of commencement of
	building works.

TAKE NOTE THAT the penalty provision below will come into effect **1 October 2023.**

94) Delay in completion of Building works applicable from 1 October 2023	Failure to complete a building within 21 months
_ applicable from 2 colorer 2020	from date of commencement of building works,
	will accrue a monthly building penalty levy equal to
	a double monthly standard levy up to and including



Revision 17 September 2022

36 months from date of commencement of
building works
Building levy penalties for non-completion of
building works and levies will be charged as follows
in term of this clause:
1.Months 21 to 36 = a monthly double levy
building penalty;
2.Months 37 to 48 = a monthly triple levy building
penalty;
3. Months 49 onwards, a monthly building penalty
as resolved by the Board of Monaghan Farm NPC
or a monthly quadruple levy building penalty until
works are completed as per the Rules and
Regulations of Monaghan Farm NPC.

RESOLUTION ADOPTED AT ANNUAL GENERAL MEETING 17 SEPTEMBER 2022 IN RESPECT OF DELAY IN COMPLETION OF BUILDING WORKS PENALTY

WHEREAS the Board wishes to apply a further punitive penalty for any delay in commencement and / or the completion of building works.

AND WHEREAS Article 11.13 of the MOI empowers the Board to impose penalty levies on Members who have failed to comply with the Title Deed obligation to build as prescribed by the Title Deed



Revision 17 September 2022

Conditions read with the Rules of MFHOA which penalty levies shall be payable in addition to the levies normally applicable to the stand owned by a Homeowner.

AND WHEREAS Homeowners and/or Developers should be mindful of the strict building regulations entrenched for the greater benefit of the Members and Monaghan Farm as a whole.

IT IS RESOLVED THAT over and above the obligations against a Homeowner entrenched in paragraph

F of the Title Deed conditions relating to the commencement of building works, a further punitive

penalty will be included in the Monaghan Farm NPC Penalty Schedule which states the following:

Failure to complete a building within 21 months from date of commencement of building works, will

accrue a monthly building penalty levy equal to a double monthly standard levy up to and including

36 months from date of commencement of building works.

A monthly building penalty levy equal to a triple monthly standard levy will be imposed for incomplete works between the months of 37 and up to and including the 48th month from date of commencement of building works.

A monthly building penalty levy as resolved by the Board of Monaghan Farm NPC will be imposed for incomplete works from month 49 from date of commencement of building works until works are complete as per the Rules and Regulations of Monaghan Farm NPC.

Consequently, building levy penalties for non-completion of building works and levies will be charged as follows in term of this clause:

- 1. Months 21 to 36 = a monthly double levy building penalty;
- 2. Months 37 to 48 = a monthly triple levy building penalty;
- 3. Months 49 onwards, a monthly building penalty as resolved by the Board of Monaghan Farm NPC **or** a monthly quadruple levy building penalty until works are completed as per the Rules and Regulations of Monaghan Farm NPC.



Revision 17 September 2022

TAKE NOTE THAT this resolution and penalty provisions will only come into effect 1 October 2023.

Schedule: version control.

1-03-2022	Final	Published to website and portal. SCHEDULE OF FEES, CHARGES AND PENALTIES NOTICE OF AVAILABILITY Ref No: 29/01/2021-792
02-2022	Amendment Procedures	Budget review process: Increment recommended by Fincom to Board of Directors. Increment principles adopted at Board meeting 1 February 2022 Final Schedule verified by Directors assigned to Fincom _ 8 February 2022
02-03-2022	Final	Published to website and portal. SCHEDULE OF FEES, CHARGES AND PENALTIES NOTICE OF AVAILABILITY Ref No: 02/03/2022 – 963
3-10-2022	FINAL	Published to website and portal. SCHEDULE OF FEES, CHARGES AND PENALTIES NOTICE OF AVAILABILITY Ref No: 03/10/2022 – 873